

Walden University ScholarWorks

Walden Dissertations and Doctoral Studies

Walden Dissertations and Doctoral Studies Collection

5-8-2024

Lived Experiences of Congolese Women with Postpartum Family Planning Counseling in the United States

Lemba Dina Nshisso Walden University

Follow this and additional works at: https://scholarworks.waldenu.edu/dissertations

Walden University

College of Health Sciences and Public Policy

This is to certify that the doctoral dissertation by

Lemba Dina Nshisso

has been found to be complete and satisfactory in all respects, and that any and all revisions required by the review committee have been made.

Review Committee

Dr. Jeanne Connors, Committee Chairperson, Public Health Faculty
Dr. Kai Stewart, Committee Member, Public Health Faculty

Chief Academic Officer and Provost Sue Subocz, Ph.D.

Walden University 2024

Abstract

Lived Experiences of Congolese Women with Postpartum Family Planning Counseling in the United States

by

Lemba Dina Nshisso

MPH, Case Western Reserve University, 2012 BS, Gardner-Webb University, 2009

Dissertation Submitted in Partial Fulfillment
of the Requirements for the Degree of
Doctor of Philosophy
Health Sciences

Walden University

May 2024

Abstract

Postpartum family planning counseling experience among immigrant women in the United States has primarily involved Latina immigrant women, with little focus on African immigrant women. To address this gap, a phenomenological study was conducted to examine lived experiences of Congolese immigrant women living in the United States (U.S.) with postpartum counseling and how this type of counseling influences their decision-making processes. Additionally, the study involved understanding Congolese immigrant women's interactions with healthcare services for family planning procurement. Data were collected from 10 Congolese immigrant women who had given birth in the U.S. and attended a postpartum family planning counseling session. Analysis of data was guided by the social cognitive theory, focusing on selfefficacy and outcome expectancy. These constructs helped to understand how experiences involving postpartum family planning counseling informed women's decision-making regarding contraceptive choices. Thematic analysis of data revealed five key themes: complexity of the U.S. healthcare system, familiarity with healthcare providers, immigrant status, perceived understanding of family planning, and factors associated with procurement. Findings of this study involve the importance of the environment in which postpartum family planning counseling takes place. Counseling experiences revealed factors that can either strengthen or undermine women's selfefficacy and outcome expectancy motivators that influence decision-making processes. This study can promote social change by empowering African immigrant women in making informed choices during family planning counseling.

Lived Experiences of Congolese Women with Postpartum Family Planning Counseling in the United States

by

Lemba Dina Nshisso

MPH, Case Western Reserve University, 2012 BS, Gardner-Webb University, 2009

Dissertation Submitted in Partial Fulfillment
of the Requirements for the Degree of
Doctor of Philosophy
Health Sciences

Walden University

May 2024

Dedication

This dissertation is dedicated to my daughters. This work is a step towards leaving my mark on the community. Knowing the purpose of my calling, however small it may be, will always be my driving force. For this reason, I urge you to trust the One who has called you, for He will equip you.

To my other half, Bob, I thank you for your dedication. You have been and will always be my rock.

To my parents, family, and in-laws, being surrounded by your support is a blessing. You have always believed in me from the moment I was born. To my dear friends, thank you for the emotional support!

To my chair, Dr. Jeanne Connors, you have given exceptional guidance throughout this journey. Thank you for your patience, expertise, and contribution in making this work one of my best.

To my grandmothers, Ana and Ana, I extend my heartfelt thanks for their prayers.

They have been a gentle bear and a fierce lion in my life.

Table of Contents

st of Tables	v
hapter 1: Introduction to the Study	1
Background of the Study	3
Problem Statement	5
Purpose of the Study	7
Research Questions	7
Theoretical Foundation	8
Nature of the Study	8
Definitions	9
Limitations and Assumptions	10
Scope and Delimitations	11
Significance of the Study	11
Summary	12
hapter 2: Literature Review	13
Literature Search Strategy	14
Theoretical Foundation	15
African Immigrants in the U.S.	16
Characteristics of Female African Immigrants in the U.S.	18
Postpartum Family Planning Counseling	19
Postpartum Family Planning Among African Immigrant Women	22
Reproductive Health Service Access and Utilization	25

Acculturation	26
Healthcare Knowledge and Use	27
Medical Trust	28
Summary and Conclusions	29
Chapter 3: Research Method	31
Research Design and Rationale	31
Role of the Researcher	31
Research Questions	32
Methodology	32
Participant Selection Logic	32
Participant Recruitment	33
Data Collection	34
Instrumentation	34
Data Analysis Plan	35
Issues of Trustworthiness	35
Credibility	36
Transferability	36
Dependability	37
Confirmability	37
Ethical Procedures	37
Summary	38
Chanter 4: Results	30

Research Questions	39
Pilot	39
Setting	40
Demographics	41
Data Collection	41
Data Analysis	42
Evidence of Trustworthiness	43
Credibility	43
Transferability	44
Dependability	44
Confirmability	44
Results	45
RQ1	45
RQ2	49
RQ3	52
Summary	54
Chapter 5: Discussion, Conclusions, and Recommendations	56
Interpretation of the Findings	57
Theme 1: Complexity of the U.S. Healthcare System	57
Theme 2: Familiarity with Healthcare Providers	58
Theme 3: Immigrant Status	59
Theme 4: Perceived Understanding of Family Planning	60

Theme 5: Factors Associated with Procurement	61
Interpretation of the Findings and SCT	63
Self-Efficacy	64
Outcome Expectancy	65
Limitations of the Study	65
Recommendations	66
Implications	68
Conclusions	69
References	70
Appendix A: Recruitment Flyer	97
Appendix B: Informed Consent	98
Appendix C: Demographic Information and Interview Ouestions	100

List of Tables

Table 1	Demographics	41
Table 2	Emerging Themes for RQ1	45
Table 3	Emerging Themes for RQ2	49
Table 4	Emerging Themes for RQ3	52

Chapter 1: Introduction to the Study

Family planning is an essential part of maternal and reproductive health (Boydell et al., 2018; Dehlendorf et al., 2018; Ewerling et al., 2018). In low-income countries, reproductive health services delivering family planning methods have served to address much more than just improving maternal health. Benefits include decreasing fertility rates to achieve population growth stability, reducing poverty through economic stability, and improving maternal education (Kadarisman, 2019; Kantorova, 2020; Tran, Yameogo et al., 2018). For this purpose, the delivery of family planning counseling ensures women receive adequate knowledge about contraceptive methods and increase their uptake. In high-income countries, delivery of family planning counseling improves maternal health by reducing unintended and short-spaced pregnancy rates (Bearak et al., 2018; Hanley et al., 2017; Ray & King, 2018).

Healthy spacing between pregnancies is associated with better maternal outcomes and quality of life. In the United States (U.S.), 40% of all pregnancies are unplanned (Ahrens et al., 2018). Unintended pregnancies are associated with poorer outcomes for mothers and children, including low birth rates (Omani-Samani et al., 2019; Pakseresht et al., 2018; Sawhill & Guyot, 2019). Unintended pregnancies also impact life in a separate way. There can be a reduced quality of life in some contexts, such as negative effects on socioeconomic status, career, and education attainment. Women have reported being willing to abort to avoid unintended pregnancy (Spierling & Shreffler, 2018).

Furthermore, due to financial or career plans, women have a desire to avoid pregnancy (Jones, 2017; Spierling & Shreffler, 2018).

Besides the impact of unintended pregnancies on life, rates of unintended pregnancy vary significantly by racial and ethnic minorities due to factors such as marital status, education, insurance, type, socioeconomic status, age at first pregnancy, and U.S.-born status (Aztlan-James et al., 2017; Hernandez et al., 2020; Jackson et al., 2017; Troutman et al., 2020). Postpartum family planning becomes a context where factors affecting the rate of unintended pregnancies among different racial and ethnic groups can be addressed, and reproductive desires can be discussed (Agbemenu et al., 2019; Agbemenu, Volpe, et al., 2018; Coleman-Minahan & Potter, 2019).

African immigrant women are among increasing ethnic or minority groups in the United States. They come from various countries, including the Democratic Republic of Congo (DRC) (United Nations High Commissioner Refugees [UNHCR], 2018; U.S. Department of State & Bureau of Population, Refugees, and Migration, 2019). Many Congolese immigrant women are of reproductive age (i.e., 15-44 years), necessitating a particular need and approach to primary care and reproductive health services (Pazol et al., 2017; Royer et al., 2020). For example, culture and tradition play a significant role in how Congolese immigrant women view health, influencing healthcare behaviors and experiences (Omenka et al., 2020). Hence, cultural competence among healthcare providers is crucial in delivering targeted approaches to reproductive health services. Previously published data showed that Congolese immigrant women's access and low utilization of reproductive health services are affected by factors such as language barriers, low health literacy, lack of understanding of healthcare systems, and lack of

health insurance coverage (Evans et al., 2022; Im & Swan, 2019; McMorrow & Saksena, 2017; Tapales et al., 2018).

Although the literature mentioned above provides some aspects of health behaviors and barriers faced by Congolese immigrant women, not enough is documented about Congolese immigrant women's experiences with reproductive health services, especially postpartum family planning counseling (Coleman-Minahan & Potter, 2019; Manzer & Bell, 2022; Pazol et al., 2017; Royer et al., 2020). The current study provided further information on postpartum family planning counseling and its experience among Congolese immigrant women living in the U.S. Chapter 1 includes the background of the study, problem statement, purpose of the study, and research questions. I also addressed the theoretical framework, nature of the study, definitions, assumptions, scope and delimitations, limitations, and significance of the study, followed by a summary.

Background of the Study

Few studies have examined the relationship between postpartum family planning and African immigrant women. Millar et al. (2017) provided information on the pattern of family planning methods and prevalence use among Somali women following their postpartum visits. Results indicated postpartum family planning counseling among Somali women was a key factor in terms of informing them about family planning methods and health service use. Another study assessed factors influencing reproductive health decision-making associated with resettlement in the United States among Somali refugee women. The study observed that Somali refugee women emphasized birth spacing rather than birth limiting methods when discussing postpartum family planning

options with healthcare providers (Agbemenu, Volpe et al., 2018). However, women had not consistently reported amicable relationships with healthcare providers during counseling sessions. Gele et al. (2020) explored factors associated with low modern contraceptive use among Somali immigrant women of reproductive age (> 18 years old) living in Oslo, Norway. The authors were interested in finding barriers to low contraceptive use because a previous investigation found that the rate of unintended pregnancies among this population was about 50% (Gele et al., 2019). The one-on-one interview revealed that women understood the importance of contraceptive use and its role in healthy pregnancy outcomes. They also reported overall satisfaction with the healthcare system. However, they felt healthcare providers had unfriendly attitudes towards understanding the Somali culture (Gele et al., 2020). Another study in the United States explored the perception of Latina immigrants during postpartum counseling and the role of physicians during counseling sessions. Participants perceived a sense of bias and stereotyping among healthcare providers when helping them decide on contraceptive methods (Carvajal et al., 2017).

Studies also highlighted that African immigrant women did not fully understand the healthcare system to make the best use. For example, 16 Congolese refugee women aged between 18 and 44 who resettled in the U.S. participated in a focus group interview to explore their experience with the U.S. healthcare system. Women noted the U.S. system was particularly confusing and faced difficulties securing appointments (McMorrow & Saksena, 2017). Another study noted a significant difference in the use of reproductive health services. Immigrant women had lower use rates at 69.0% compared

to U.S.-born women at 73.0% (Tapales et al., 2018). Potter et al. (2017) explored differences in postpartum family planning policies in the U.S. and Mexico's healthcare systems. The authors observed that U.S. healthcare policies involving family planning methods a significant barrier for immigrant women to meet their contraceptive needs.

Problem Statement

The postpartum period is crucial for initiating family planning, also referred to as contraception, among women of reproductive age. Counseling that is offered during the postpartum period allows women to make informed decisions about family planning methods (Brunson et al., 2017; Cooper & Cameron, 2018). Contraception reduces the number of unintended pregnancies, which results in short pregnancy intervals and low pregnancy risks (Sawhill & Guyot, 2019). In the U.S., unintended pregnancies account for 40% of all pregnancies, and 36% of those pregnancies have interpregnancy intervals that are less than 18 months in 2014 among women between 15 and 44 (Ahrens et al., 2018; Kavanaugh & Jerman, 2018). Among unintended pregnancies, disparities exist between immigrant and U.S.-born women. Non-U.S.-born women account for 73% of unintended pregnancies compared to U.S.-born women (Kim et al., 2016). Agbemenu et al. (2022) and Omenka et al. (2020) argued that studies on postpartum family planning have often classified immigrant women into one single group of foreign-born to compare use with U.S.-born women. Therefore, strategic family planning counseling recommendations are not always tailored to different immigrant populations (Dehlendorf et al., 2017; Ngendahimana et al., 2021). The U.S. is home to immigrants from various countries, including the DRC. Data from the DRC showed that the total fertility rate in

the DRC was 6.5 children, with contraceptive use of 11.2% in 2019 (Nyoni et al., 2021). In recent years, heightened involvement from the Congolese government has led to family planning initiatives to assess attitudes and beliefs, increase counseling, and address barriers to contraception use (Ho & Wheeler, 2018; Tran, Gaffield, et al., 2018; Tran et al., 2020; Zivich et al., 2019). However, with resettlement in the U.S., it is not known to what extent Congolese women's attitudes and beliefs affect their experience with postpartum family planning counseling offered in a new country.

Studies have begun to evaluate contraceptive behaviors and experiences of different immigrant groups such as Latinas or Somali women (Agbemenu, Volpe, et al., 2018; Carvajal et al., 2017). While attitudes and experiences of postpartum family planning counseling in the literature have changed among Latinas, more studies remain to be evaluated to explore the postpartum family planning counseling experiences among different African immigrant populations. Agbemenu, Volpe, et al. (2018) and Millar et al. (2017) evaluated patterns of postpartum contraceptive use and the decision-making process of reproductive health care among Somali women and found that of the 88.3% of women who received counseling, 80% chose a form of contraception. These studies have begun to shed light on understanding contraceptive behaviors among African women immigrants. Still, they only focused on reporting the prevalence of use among Somali women and their overall experience with reproductive health services. In 2017, the U.S. Department of Homeland Security (2018) reported 8,709 Congolese received the status of lawful permanent residents, a number that increased from 5,345 in 2015. Also, in 2017, an additional 8,343 Congolese were admitted under non-immigration status (U.S.

Department of Homeland Security, 2018). As Congolese women of reproductive age resettling in different states are increasing and given a high fertility rate from the native country, it is essential to understand how to provide tailored family planning counseling. Therefore, it is vital to learn how different African immigrant women communities perceive the importance of postpartum family planning and their lived experiences of postpartum family planning counseling in the U.S.

Purpose of the Study

The purpose of the study was to examine lived experiences of Congolese immigrant women living in the U.S. with postpartum counseling and explore how counseling informs their decision-making processes. This study also involved understanding their interactions with healthcare services for family planning procurement. Qualitative literature on attitudes of African immigrant women living in the United States on counseling and modern contraceptive behavior following the postpartum period is limited. Showers (2018) claimed studies on Black immigrants had been limited to shield them from discrimination or marginalization. However, as part of increasing diversity in immigrant populations, research should evaluate how Congolese women deal with various health behaviors and assess how public health initiatives improve health outcomes.

Research Questions

RQ1: How do Congolese immigrants perceive their experiences with postpartum family planning counseling?

RQ2: How do postpartum family planning counseling experiences inform their decision-making processes?

RQ3: How do Congolese immigrants interact with healthcare systems for the procurement of modern family planning methods?

Theoretical Foundation

The theoretical framework for this study was the social cognitive theory (SCT). The SCT is used to explain interplays between environmental and personal factors that fundamentally affect human behaviors through reciprocal determinism. The theory is used to understand how individual experiences, actions and behaviors, and environmental factors influence health behaviors (Bandura, 1986; Fertman & Allensworth, 2016, p. 65; Glanz et al., 2015, p. 171). Bandura (1986) claimed constructs of the SCT include environment, situations, behavioral capacity, outcome expectations, outcome expectancies, self-control, observational learning, self-efficacy, and emotional coping. By delving into the shared experiences of Congolese women, I used the SCT to identify factors that could either strengthen or undermine women's self-efficacy and outcome expectancy motivators. These factors play a crucial role in influencing decision-making processes during family planning counseling. The SCT was also used to provide a framework for interview questions, data collection, and explanation of a phenomenon or lived experience (Creswell, 2014, p.64).

Nature of the Study

The purpose was to conduct a qualitative study using a phenomenological approach to examine perceptions of Congolese immigrant women receiving postpartum

family planning counseling and their experiences with health services. Phenomenology is a way to investigate and obtain information pertaining to behavior based on personal values, environmental contexts, and cultural factors that inform decisions about contraceptive use (Downey et al., 2017). I used one-on-one semi-structured interviews. Participants were asked to describe their lived experiences when receiving family planning counseling during the postpartum period and explain how information they received informed their decisions about choosing a method. The phenomenological approach was used to understand experiences of going through postpartum family planning counseling, factors influencing their decision-making process, and barriers and enablers to postpartum contraceptive access.

Definitions

In this section, I define and elaborate on key terms and phrases.

African Immigrant Women: Women who were born in any African country and migrated to the U.S.

Congolese Woman: Woman who was born in the DRC.

Family Planning methods: Family planning and contraceptive methods are used interchangeably. It consists of methods to avoid unintended pregnancies, space births, or limit births. Methods include pills, implants, injectables, the patch, vaginal rings, intrauterine devices (IUDs), condoms, sterilization, Lactational Amenorrhea, fertility awareness, and withdrawal (Cooper & Cameron, 2018; World Health Organization [WHO], 2018).

Postpartum Counseling: Sessions which consist of educating women about postpartum family planning methods (e.g., how each method works, advantages and disadvantages of each method, and costs) and understanding their needs regarding future pregnancies. While family planning counseling can also be done during prenatal visits, it is standard care during the postpartum period (Schivone & Glish, 2017).

Postpartum Period: This is separated into three different periods. The first is the acute period, which is the first 12 hours post-delivery. The second stage, the subacute period, is between 2 and 6 weeks. The final stage is the delayed postpartum period and lasts up to 6 months (Romano et al., 2010).

Reproductive Health: The WHO (2010) defined it as "a state of complete physical, mental and social well-being and not merely the absence of disease or infirmity" (p. 10). Reproductive health involves all matters of the reproductive system and includes education and information about sexual health and family planning as well as sexual desire (Metusela et al., 2017; WHO, 2010, 2017).

Limitations and Assumptions

One potential limitation was social desirability bias, where participants tend to answer questions based on what they felt was socially acceptable. One-on-one interviews were meant to address this limitation by allowing participants to answer truthfully without being influenced by someone else's responses (Bergen & Labonté, 2020). The assumption was that participants shared their experiences voluntarily and provided honest answers to interview questions.

Scope and Delimitations

The study involved the postpartum period and family planning counseling that women received to decide on choice of contraceptives. The study was delimited to Congolese women who immigrated from the DRC and lived in the U.S. Participants resided in North Carolina. Since this study was specific to Congolese women, it added to the body of research concerning African women immigrants and their reproductive health decisions while living in America. However, results of this study cannot be transferred to other African immigrant women groups.

Significance of the Study

Implications for social change include valuable knowledge for program developers, reproductive health counselors, healthcare providers, and researchers involving improving reproductive health needs among Congolese immigrant women.

Long-term results of programs tailored to different African immigrant populations would hopefully increase knowledge, increase rates of family planning use, decrease unplanned pregnancies, and improve healthcare service delivery. Study results will contribute to the community and institutions by understanding how Congolese immigrant women perceive family planning counseling and navigate the healthcare system for procurement.

Understanding factors that influence their decision-making processes is essential for interventions and interaction between healthcare providers and patients. Furthermore, when providers are informed about different immigrant populations and barriers they face, services offered can be less subject to assumptions and biases (Agbemenu, Hannan et al., 2018; Okeke-Ihejirika et al., 2018).

Summary

Reproductive health services are underused among immigrant women. Several challenges, such as limited knowledge of the healthcare system, resettlement challenges, and language barriers in the host country, are reasons why foreign women place low prioritization on reproductive health services (Åkerman et al., 2019; Mehta et al., 2018; Metusela et al., 2017). I focused on the postpartum period and experiences with postpartum family planning counseling among Congolese immigrant women. Chapter 1 included an overview of the problem, research methods and theoretical foundation, and social change implications.

Chapter 2 includes in-depth research of evidence, and Chapter 3 contains a review of the methodology. Chapter 4 includes study results, and Chapter 5 includes findings from results as well as implications for social change.

Chapter 2: Literature Review

The purpose of the study was to examine lived experiences of Congolese immigrant women living in the U.S. with postpartum counseling and explore how counseling informs their decision-making processes. This study also involved understanding this population's interactions with healthcare services for family planning procurement. Immigrant populations and minorities are disproportionally affected by reproductive healthcare services and outcomes. Specific services such as postpartum contraceptive counseling among immigrant women have not been widely used due to barriers such as language, lack of proper insurance coverage, not understanding the healthcare system, and lack of trust in healthcare providers (Mehta et al., 2018; Ngendahimana et al., 2021; Oakley et al., 2018).

Studies among foreign-born women living in the United States have been increasingly focused on Latinas (Carvajal et al., 2017; Hernandez et al., 2020; Oakley et al., 2018; Tapales et al., 2018). As a result, policy interventions to educate immigrant women have essentially been based on experiences of a specific population (Omenka et al., 2020). In contrast, subpopulations such as women from sub-Saharan Africa have been scarcely studied. For example, satisfaction with postpartum contraceptive counseling among African-born immigrant mothers and how they inform contraceptive use has not been extensively studied. Despite the increase of African women immigrating to America, scarcity of research among African immigrants and postpartum counseling is evident, so addressing the gap in the literature was significant (Agbemenu et al., 2022; Omenka et al., 2020). Congolese immigrant women were chosen for this study to

examine their attitudes and experience with postpartum counseling and understand how they have impacted their decision-making processes.

The literature review drew on studies that explored experiences, attitudes, and knowledge about postpartum contraceptive counseling among immigrant women.

Immigrant populations such as Latinas, Ethiopian and Somali women have gained more attention in research to understand how policy interventions impacted them differently (Hernandez et al., 2020; Oakley et al., 2018; Oliphant, 2019). Early literature on African immigrant women, including Congolese-born women, has primarily focused on refugees and asylum seekers (Im & Swan, 2019; McMorrow & Saksena, 2017; Mehta et al., 2018; Royer et al., 2020; Saksena & McMorrow, 2017). Because of their different immigration status and experiences involving displacement and resettlement, these minority groups are more likely to report different experiences in the U.S. (Mehta et al., 2018; Royer et al., 2020).

Literature Search Strategy

I used a variety of databases to acquire articles for the literature review. The following search engines and databases were used: EBSCOHost, PubMed, Medline Plus, Google Scholar, and Walden Library. The first step was to find relevant literature on immigrant women and postpartum contraception. I then narrowed my search to include African immigrant women to identify current findings and the gap in literature. I used the following terms: *immigrant women*, *African immigrant women*, *postpartum family planning counseling*, *postpartum contraceptive counseling*, *family planning knowledge*, *postpartum education*, *reproductive health services*, *gynecologic care*, *social cognitive*

theory, affordable care act, and preventive services. Since there was little information about Congolese-born women settled in the U.S, I mostly referred to studies that used immigrant women from other African countries and other subgroup populations that shared similar characteristics.

Due to the nature of the topic, I chose articles that were published between 2017 and 2022. Sources involving the theoretical framework and historical information were not limited to the five-year review.

Theoretical Foundation

I examined experiences of Congolese immigrant women using the SCT.

Behaviors are learned within a social context through observational learning and the presence of reciprocal interactions between personal and environmental behaviors (Chess et al., 1978; Wood & Bandura, 1989). The SCT is used to understand how thoughts and emotions affect individual decisions involving new health behaviors. The SCT is also used to address the importance of experience as a determinant of individual behaviors, as well as environment and social support. In making decisions about new health behaviors, the SCT highlights the importance of the individual having beliefs and capabilities as well as the knowledge about desired outcomes before engaging (Chess et al., 1978; Lopez et al., 2016). Thus, for this study, I focused on self-efficacy and outcome expectancy constructs.

Self-efficacy refers to an individual's belief in their capacity to exert control over their goals and expectations, as well as their ability to carry out the necessary actions to achieve the desired outcomes. This construct, as highlighted by Bandura (2004) and

Chess et al. (1978), plays a crucial role in predicting not only the adoption but also sustainability of new health behaviors. Expectancy outcome construct, on the other hand, pertains to the individual's confidence in their ability to engage in new health behaviors based on the anticipated outcomes. Subsequently, the purposefulness and meaningfulness of the new task serve as powerful motivators for the individual (Chess et al., 1978; Lopez et al., 2016).

The adoption of new health behaviors, such as contraception and uptake, requires individuals to possess adequate knowledge to make informed decisions.

Therefore, postpartum family planning counseling acts as a medium for women to gain insights into different family planning methods. These counseling sessions are pivotal in enabling women to make decisions that resonate with their reproductive health desires.

Consequently, the experiences encountered during these counseling sessions form part of the cognitive process that will shape women's decisions, as elucidated by the SCT. By delving into the shared experiences of Congolese women, I examined factors that could either strengthen or undermine women's self-efficacy and outcome expectancy motivators during family planning counseling. By identifying these factors, efforts can be directed towards establishing a supportive and empowering environment for Congolese women.

African Immigrants in the U.S.

In the last decades, the U.S. has seen increased immigrants from African countries (López & Bialik, 2017). In the early 1990s, most African immigrants came from Anglophone-speaking countries such as Ghana, Nigeria, Sierra Leone, and Liberia (Tamir, 2022). In contrast, most asylum seekers and refugees came from non-English

speaking places such as Sudan, Ethiopia, and the DRC (Anderson, 2017). Over the years, African-born immigrants have come for reasons such as diversity visas, family resettlement programs, educational purposes, international schooling, and refugee and asylum programs (Anderson, 2017; Refugee Processing Center, 2022). In the U.S., African immigrants have increased by more than 41% from 2000 to 2013 (Anderson, 2017).

States such as New York, Maryland, and Minnesota have been the primary destination for immigrants and hold about 12% of all African immigrants (Opoku-Dapaah, 2017; Tamir, 2022). Increasingly, states like Texas, Georgia, Illinois, and North Carolina have become primary locations (Opoku-Dapaah, 2017). According to the Refugee Processing Center (2017), North Carolina became the third largest resettlement location in 2017, while Congolese immigrants have become the largest immigrant group coming to North Carolina.

Reasons for immigration among sub-Saharan populations have been well documented (Chand, 2019; Covington-Ward et al., 2018; Echeverria-Estrada & Batalova, 2019). For many Congolese immigrants, one of the ultimate goals has been to achieve economic advancement to offer a better future and stability for their families, just as documented in the literature for many of their African counterparts (Anderson, 2017; Chand, 2019; Covington-Ward et al., 2018). Although early literature had focused on African men's immigration and their experience in the United States, growing data showed a surge in women's immigration (Muruthi et al., 2017; Showers, 2018). The

number of female African immigrants has been rising in the last decades compared to male African Immigrants (Nawyn & Park, 2019a; Semu, 2020)..

Characteristics of Female African Immigrants in the U.S.

Immigrant women have unique needs and play an essential role in their host country (Nawyn & Park, 2019; Okeke-Ihejirika et al., 2018). Muruthi et al. (2017) documented that one of the roles played by African women immigrants was their increased financial contribution to their families in their native countries. In their study, women participants were from Rwanda and Kenya, were between 22-30 years old, and had lived in the United States for 1 to 6 years (Muruthi et al., 2017). Participants had a 60-120 min interview in English. Themes developed from the study revealed first that women felt a sense of responsibility to send money back to their families. Second, participants' families had a skewed understanding of financial expectations, and finally, participants wanted to invest in their country of origin. In parallel, another study examined the immigration experience of 14 Ethiopian women living in Washington, D.C. The major themes discussed in the findings were: (1) arrival at the airport, (2) living arrangements, (3) finding jobs, and (4) furthering education. Participants discussed the importance of saving money to send it back to their native country. Women also expressed the importance of relying on institutions like family and friends to help them navigate their new life in the United States (Oliphant, 2019). While these two studies do not address Congolese women specifically, it is essential to note the additional role African women have taken on while living in the U.S. and how it impacts their decisionmaking experience and process.

Aside from their vital roles, research has also documented African women's needs in their host countries. For example, one study focused on Congolese refugee women to explore their resettlement experiences in the United States. Women described experiencing loneliness, isolation, and being overwhelmed trying to juggle professional and family life in their new country (Saksena & McMorrow, 2019). Similarly, McMorrow and Saksena (2017) conducted an earlier study to assess the health needs and look at the assets of Congolese refugee women. Women reported that the social support was tailored around their community with people from back home and to make a connection with American Society. Health needs and frustration included their lack of understanding of the healthcare system, precisely the meaning of medical appointments and understanding of the medical insurance system (McMorrow & Saksena, 2017). These studies highlighted women immigrants' position in the United States, psychosocial changes in adapting to a new country that influence their everyday living and eventually impact how they make decisions. Among African immigrant women's priorities of making a new life in their new country, little attention has been devoted to reproductive health and their experiences with navigating the system, thus prompting further attention in research and public health.

Postpartum Family Planning Counseling

Postpartum is a crucial time for education about family planning. Reasons include a woman's desire and decision about future pregnancies, cost-effectiveness, and a higher chance of continuing contraception use (Williams et al., 2019). A woman can regain fertility within six weeks if she is not fully breastfeeding during the postpartum period.

During this time, a woman runs into a higher risk of unintended pregnancy and short-spaced pregnancy interval. For women who choose to breastfeed exclusively, the literature has shown that lactational amenorrhea provide at least 6 months of protection against pregnancies (Cooper & Cameron, 2018; Glasier et al., 2019; Sridhar & Salcedo, 2017). Healthcare providers play a critical role in interacting with women and discussing further intentions on pregnancies (Manzer & Bell, 2022; Thiel de Bocanegra et al., 2020). Due to different contraceptive methods, postpartum family planning counseling is crucial in determining the decision to use contraceptives and the best timing for each method (Cooper & Cameron, 2018). Postpartum counseling remains an underused option among providers and patients (Moniz et al., 2017; Ray & King, 2018).

Before exploring the topic of postpartum family planning, I looked at differences in unintended pregnancy, preterm births, and birth spacing among women across races and ethnicities. Some studies have shown that race and ethnicity were associated with unintended pregnancies. Quinn et al. (2020) did a cross-sectional study from a national sample of 2302 women to examine patterns of unintended pregnancies and contraceptive use among different racial and ethnic minority groups of women veterans. In the sample, 57.2% of pregnancies were unintended. Non-Hispanic Black women showed the highest percentage (66.8%) of unintended pregnancies, followed by Hispanic and non-Hispanic women, respectively (60% and 50.8%). The authors found that factors associated with unintended pregnancy were low contraception knowledge and reproductive health service use among minorities. The study's limitation is that race and ethnicity categories were limited to White, non-Black, and Hispanic women. Moreover, participants were veteran

women and VA primary care users, which meant that most women were from racial and ethnic minority groups and had poorer overall health (Quinn et al., 2020).

Studies in the United States have shown that women of racial and ethnic groups compared to white non-Hispanics had a higher rate of preterm births (Tucker et al., 2020). Oliver et al. (2018) conducted a study of four different groups of women to compare preterm birth rates and gestational age. First, race and ethnic group were categorized into four sections: U.S.- born non-Hispanic white, U.S.- born non-Hispanic black, African-born black (women from west African countries), and Somali-born. Preterm was classified as gestational age less than 36 weeks, full-term was between 37-40 weeks, and later-term was 41 weeks. Interestingly, the authors found that Somali-born women had a lower rate of preterm birth (5.9%) compared to U.S.-born non-Hispanic black (13.0%), African-born black (8.4%), and U.S.-born non-Hispanic whites (7.9%). However, Somali-born tended to have longer gestational age (41 weeks) (Oliver et al., 2018). One of the study's strengths was that participants were drawn from the same registry, indicating that obstetric practice guidelines were uniform which allowed the assessment of different ethnic groups. Another study found that U.S.-born black women had a higher preterm birth rate than foreign-born black women and U.S.-born white women (11.1% vs. 7.9% vs. 6.8%, respectively). One of the primary explanations for this persistent disparity between African-born women and U.S.-born blacks is the health profile. African Americans tended to have comorbidity factors such as hypertension and smoking and social factors such as lower educational attainment and less likely to be married (DeSisto et al., 2018).

Recent research examining further the difference between African American women and foreign-born women continued to point out that foreign women had a lower rate of preterm births but had a higher percentage of short pregnancy intervals than U.S.-born women (Agbemenu et al., 2019). Short pregnancy intervals are associated with the risk of preterm births when interpregnancy intervals are close together, not allowing a women's body to recover properly. The difference in pregnancy spacing has also been noted among women of different socioeconomic backgrounds. Cross-Barnet et al. (2018) conducted a study with Medicaid participants to explore facilitators and barriers to healthy pregnancy spacing. Type of medical insurance coverage, which is sometimes used as a proxy for socioeconomic status, indicated that women with Medicaid were less likely to go for postpartum visits. This was a barrier because women were less likely to receive optimal information regarding family planning and the importance of birth spacing (Cross-Barnet et al., 2018).

Postpartum Family Planning Among African Immigrant Women

To further understand the experience with postpartum family planning counseling, we first explored studies that discussed African immigrant women's knowledge, attitudes about family planning, and then satisfaction with family planning methods. These concepts are explained to show how they are related to the study's topic and identify gaps in the literature review. Due to limited information on Congolese immigrant women and their use of reproductive health, this section drew on studies assessing these factors among different African immigrant women groups living in the United States.

Knowledge. The level of knowledge about family planning methods among African women has been crucial to evaluating factors influencing family planning and the decision-making process. For example, 30 refugee and asylum Somali women 18 years or older were recruited for a focused group interview in a qualitative study. Some women reported high knowledge of birth control options and differences between them based on a previous experience (Agbemenu, Volpe, et al., 2018). A quantitative study by Millar et al. (2017) supported the argument of the relationship between knowledge and contraceptive uptake. In their study, 325 women were included in the analyses to evaluate contraceptive use patterns among Somali immigrant women following their postpartum period. Participants had to have had at least one delivery, and they received family planning counseling within 90 days of delivery (Millar et al., 2017). However, a high knowledge of family planning methods does not always translate to high use. A crosssectional revealed that knowledge about family planning was relatively high. Out of 100 participants, only 3.4% reported having no knowledge of family planning methods. While 64% of the mothers reported a desire for future pregnancy, only 35% of the sample were currently using a family planning method (Agbemenu et al., 2020).

Attitudes. The concept of postpartum family planning and understanding the critical time for uptake plays a role in the way women decide to seek counseling. Hence, attitudes towards family planning methods have been a factor in assessing views and use. In a qualitative study of Somali and Congolese refugee women, the concept around the role of family planning varied according to their country of origin and resettlement in the United States. Somali women emphasized that women should not manipulate their fertility. In

some instances, women mentioned that it was acceptable to use contraception but only for birth spacing. On the other hand, Congolese women were accepting of the idea of birth spacing in addition to limiting total births (Royer et al., 2020).

The response from Somali women has been previously supported by another qualitative study by Egeh et al. (2019) with Somali religious leaders. Seventeen religious leaders, mainly from the Islamic faith, between 28-59 years, were recruited for an indepth interview to discuss their views on birth spacing. The leaders' main view was that family planning methods were acceptable as long as the goal was to space births. Leaders stressed that the idea of limiting births is a sin. Furthermore, participants added that not all family planning methods were acceptable, even in the case of birth spacing. They encouraged couples to limit to breastfeeding or contraceptive pills as a tolerable method (Egeh et al., 2019).

Another study using a cross-sectional method was conducted among women refugees resettled in Buffalo, NY. Participants were between 18-50 and had at least one child or were currently pregnant. Most women came from the DRC and Somalia, 49% and 30%, respectively. When asked about attitudes around family planning, almost 25% of women reported that becoming pregnant was a timing decided by God, but 26 % had a time frame in mind for future pregnancies, the remaining 36% chose to either not answer or did not want a future pregnancy (Agbemenu et al., 2020).

When it came to attitudes towards postpartum family planning, family and a partner's role occupied an essential role in women's final decision about use. Royer et al. (2020) explored two refugee groups 'attitudes, Somali and Congolese women between 18

to 64 years old. The authors documented the nuances between the two study groups when assessing themes around reproductive health access and barriers, family planning, and the decision-making process. As part of the decision-making process, Somali women reported not needing to discuss family planning with family or partners. On the other hand, Congolese women had mentioned that their husbands had the final say about family planning methods, and marriage could end if the wife were to go against that decision. Some Congolese pointed out that the reason for not currently using any methods was because their spouses were not in the U.S. (Royer et al., 2020).

Satisfaction. Satisfaction is an essential indicator of whether there has been previous use of family planning methods and determines how likely a woman is to decide on a method based on experience from someone in her social environment. An example of this can be found in the study by Agbemenu, Volpe, et al.(2018) of Somali immigrant women living in Minnesota. Participants revealed that they were more likely to find the benefits of family planning based on sharing personal experiences. Women had shared that their choices were sometimes based on advice from other women (Agbemenu, Volpe, et al., 2018).

Reproductive Health Service Access and Utilization

Previous studies have shown the extent of disparities between immigrants and U.S.-born women regarding reproductive health services utilization and preventive health services (Tapales et al., 2018; Troutman et al., 2020). Furthermore, data showed that postpartum visits have a low utilization rate (Millar et al., 2017). Some of the factors influencing such a low rate included a level of acculturation, lack of knowledge about the

healthcare system, cultural beliefs, and even variance in counseling practices (Dehlendorf et al., 2017; Gele et al., 2020; Ngendahimana et al., 2021; Pinter et al., 2017; Thiel de Bocanegra et al., 2020).

Acculturation

According to Berry's (2005) definition, acculturation is the process of integrating a new culture into one's own and how individuals respond to the process. Social constructs that influence acculturation include language, religion, age at immigration, type of immigration, length of stay in the host country, and education and socioeconomic status (Berry, 2005; Fox et al., 2017). African immigrants have mentioned language as a barrier in navigating the healthcare system. A qualitative analysis of African refugees and immigrants revealed that their no or low English proficiency hindered interaction with (Lynn Woodgate et al., 2017) healthcare providers. Participants were from 15 African countries, including the DRC, Ethiopia, and Somalia. One female participant mentioned that she would just nod at the doctor but left the facility clueless. In some healthcare settings where an interpreter was present, there was a change in health behaviors and an increased use of healthcare services (Ejike et al., 2020).

The type of immigration, such as refugee or asylum and socioeconomic status, can impact reproductive health outcomes due to low healthcare utilization. Agbemenu et al. (2019) conducted a study to test if lower SES, risk factors associated with the refugee process, and low familiarity with the healthcare system were associated with poorer reproductive health outcomes among African refugee women in comparison to U.S.-born

women. African women participants were mainly from Burundi, the DRC, Eritrea, Rwanda, and Somalia. Medicaid status was used as a proxy for SES, and maternal country of birth was used as a proxy for refugee status. African refugee women represented 1%, while U.S.-born blacks and Whites represented 22.5% and 76.5%, respectively. Interestingly, the results showed a positive trend with African immigrants having overall better reproductive health outcomes, such as lower rates of preterm births. However, women reported late perinatal care and low reproductive care utilization compared to U.S.-born women (Agbemenu et al., 2019). This study was limited because the country of birth used as a proxy did not properly differentiate the type of immigration status (e.g., refugee, asylum, and students). The representation of African refugees was reasonably low compared to U.S.-born women. The concept of acculturation is an exciting topic that can be further studied in future research.

Healthcare Knowledge and Use

Health concerns, access to health, and health insurance coverage are areas of interest where public health can address the gap of inequalities by understanding the health experiences of Congolese immigrants. Saksena and McMorrow (2019) highlighted in their qualitative study that Congolese refugee women knew the importance of having health insurance but did not fully understand how to use it. For example, women mentioned having Medicaid coverage and could not comprehend why they would still get a bill from the hospital. Participants also had difficulties familiarizing themselves with billing, payments, and copay processes (Saksena & McMorrow, 2019). Until recently, the focus has been on women facing challenges in their reproductive health needs. In

Massachusetts, a study was conducted on Congolese and Somali women to understand the low reproductive health service utilization. The authors looked at gynecological seeking behavior and discussed barriers to health access. Gynecological care was sought only for pregnancy or when something wrong was causing extreme pain (Mehta et al., 2018).

Medical Trust

Another factor influencing health service utilization is satisfaction with health care providers. If healthcare providers are not culturally sensitive, immigrant women will be less likely to seek care or talk about family planning intentions. Some studies have shown that women valued their healthcare providers' advice when women felt like family planning counseling was individualized and tailored to their current needs and expectations. Women expressed the desire to be heard and trusted that they would make the right choice based on providers' recommendations (Sundstrom et al., 2018). In their qualitative analysis, Cross-Barnet et al. (2018) reported that women sometimes felt pressured to choose a family planning method without being explained or aware of their side effects or contraindications adequately. Participants also said that some providers did not allow them time to think and decide on a method during their postpartum visit (Cross-Barnet et al., 2018). Mistrust also includes the fact that healthcare providers can be prejudiced, assuming that certain ethnic groups would only choose one method over another. For example, Latina women have expressed a feeling of being unheard and just presented with certain family planning options during counseling. Some also mentioned

that providers were trying to rush them into deciding (Carvajal et al., 2017; Rosenthal & Lobel, 2020).

Summary and Conclusions

This chapter included current research on postpartum family planning counseling, African immigrant women, and reproductive healthcare service use and satisfaction. Postpartum family planning counseling is not well-discussed among African immigrant communities, especially Congolese women. Understanding experiences with counseling among women will empower them and make them feel open about discussing family planning desires and options. Additionally, it will give healthcare providers tools for understanding the uniqueness of immigrant groups.

Postpartum counseling increases contraceptive use and decreases pregnancyrelated health risks to mothers and babies, including unintended pregnancies and preterm births (Thiel de Bocanegra et al., 2020). Female immigrants have received little family planning counseling due to acculturation factors like language barriers, lack of understanding of the healthcare system, and mistrust of healthcare providers.

Participants in this qualitative study provided valuable contributions to the body of research involving Congolese women. Literature on African Immigrant women is focused on African refugee women. For this reason, not all of experiences can be generalized to immigrant populations. Post-resettlement due to war conflicts and displacement have been shown to affect women differently. However, these studies provided baseline insights regarding some factors related to reproductive health experiences among African immigrant women. Furthermore, I identified the gap in

research. The next step was to study experiences of nonrefugee women and explore their experiences with postpartum family planning counseling.

The postpartum period is vital to help women decide about family planning during their reproductive years. My goal was to provide in-depth information from Congolese immigrant women for healthcare providers and policymakers. Chapter 3 includes an explanation of the methodology and process of finding women as well as data to understand the phenomenon.

Chapter 3: Research Method

The purpose of the study was to examine lived experiences of Congolese immigrant women living in the U.S. with postpartum counseling and explore how counseling informs their decision-making processes. This study also involved understanding Congolese immigrant women's interactions with healthcare services for family planning procurement. This chapter includes information about the methodology that I used to explore this topic. I explained the rationale for the research design. I discussed how participants were selected, ethical implications, setting, and my role as the researcher. Additionally, I explained data collection, analysis, and management.

Research Design and Rationale

Phenomenology was the most appropriate approach for a comprehensive data analysis. In-depth interviews were used with semi-structured questions. I used a set of prepared questions but let conversations flow according to participants. Questions were written to drive conversations and remain on topic.

Role of the Researcher

My role as a researcher was to provide an environment that was suitable to conduct each interview. Since interviews were semi-structured, my role was to ask questions that helped participants be comfortable telling their stories and open up about their experiences. One of my roles as the researcher was determining when data had reached saturation. The sampling methods helped reduce the researcher's bias, which can occur with purposive sampling, where the researcher is looking for information-rich

subjects. However, the snowball method helped reduce the researcher's bias (Clark & Vealé, 2018).

Research Questions

RQ1: How do Congolese immigrants perceive their experiences with postpartum family planning counseling?

RQ2: How do postpartum family planning counseling experiences inform their decision-making processes?

RQ3: How do Congolese immigrants interact with the healthcare system for procurement of modern family planning methods?

Methodology

Participant Selection Logic

The study population was Congolese immigrant women born in the DRC who relocated permanently to the U.S. and have resided for at least 5 years. Participants were between 18 and 45 and had at least one child born in the U.S. within the last 5 years. I excluded women who had refugee or asylum status. I collected information that was pertinent to the study during individual in-depth interviews. I recruited 10 participants, after which I reached saturation.

I used purposive and snowball sampling methods. Purposive sampling involves looking for information-rich participants and providing data to enrich the research topic (Bernard, 2017). One of the disadvantages of using purposive sampling is judgment bias. One way to reduce such bias is to use the snowball sampling method to access subjects using existing participants who are already enrolled in the study (Naderifar et al., 2017).

This allows access to hard-to-reach populations. This study's subjects were immigrant women who needed to talk about their healthcare providers' experiences. If they were approached by acquaintances, they were more likely to accept to become part of the research.

Participant Recruitment

I recruited participants from counties in North Carolina. The state is home to many African associations, including the Congolese Community Association. I posted flyers (see Appendix A) around African beauty salons, African churches, and international grocery stores. I contacted owners of these establishments to help spread the word to their female customers and contacted leaders of African community groups. Approaching community leaders and organizations as points of contact to recruit participants was more successful than using phone calls, email, or flyer strategies alone (Agbemenu, Hannan et al., 2018; Olukotun & Mkandawire-Valhmu, 2020). Flyers contained a summary of research, inclusion criteria for participants, and my contact information. I conducted a phone interview with interested participants and scheduled an individual interview. Participants were legal immigrants in the U.S. and at least 18 (and had a valid state-issued identification card), had lived in the U.S. for at least five years, had a child born in the U.S. within the last five years, received postpartum family counseling at a hospital following a birth or follow-up postpartum visit, sought or tried to navigate the healthcare system for contraception procurement, spoke English, and signed consent forms before interviews.

Before posting flyers, I submitted my application to the Institutional Review Board (IRB) to collect data after receiving approval to work with human subjects.

Data Collection

I collected data through one-on-one in-depth interviews while asking semi-structured questions. The process started once I received approval from the IRB (see Appendix B). Before interviews, I asked participants if they had questions about the study. Then, participants signed informed consent forms or verbally consented. Each interview lasted between 40 and 45 minutes, and participants had the option to choose where meetings took place. Options included a public library, church, and online. I used questions to begin interviews and let participants feel at ease using a conversational manner. At the start of interviews, I also asked participants for their permission to audio record. If they declined to be recorded, I informed them that I would take notes and type them shortly after interviews.

Instrumentation

Data collection instruments included an audio recorder and interview question sheets (Barrett & Twycross, 2018). Demographic information (Appendix C) was collected at the beginning of the interview. Interview questions (Appendix C) were developed according to the researcher-developed instrument using literature sources on contraceptive counseling perception, contraceptive decision making, and phenomenological inquiries (Gele et al., 2020; Metusela et al., 2017; Oakley et al., 2018; Royer et al., 2020). The next step was to test the researcher-developed instrument to assess if the interview questions would capture what they were designed to measure. One

process used in qualitative research is doing a pilot study, which helps determine the study's feasibility, participant selection plan, data collection, and data analysis method (Ismail et al., 2017). I performed a pilot study once the IRB was approved (#09-21-22-0420219).

Data Analysis Plan

Data management is essential for qualitative analysis. The amount of data produced from the interviews needed transcribing, coding, and sorting. Unlike quantitative analysis, qualitative research does not have dedicated software to perform the coding process. However, software analysis tools are available to store the data and support the researcher in managing the data and coding. I used NVivo to complete the data analysis and find emerging themes that helped form the meaning of participants' lived experiences.

Issues of Trustworthiness

Qualitative analysis requires a certain level of replicability of the process and the results. The researcher ensures validity through accuracy and reliability through consistency within the qualitative approach. Trustworthiness refers to a set of criteria that the inquirer uses to determine the quality of the study and the confidence that the reader will have in the results (Cypress, 2017; Guba & Lincoln, 1982). Traditionally, the criteria were identified as truth value, applicability, consistency, and neutrality (Guba, 1981). They have evolved to more practical terms of credibility, transferability, dependability, and confirmability (Guba & Lincoln, 1982). In this section, I described strategies I

applied throughout the methodology to achieve criteria established by Guba and Lincoln (1982) to confirm the trustworthiness of the research.

Credibility

Analogous to internal validity, credibility seeks to ask whether the findings are believable. Among the methods discussed in the literature, I focused on member checks to ensure that participants' lived experiences were kept throughout the interpretation (Cypress, 2017; Guba & Lincoln, 1982; Johnson et al., 2020; Korstjens & Moser, 2018). I continuously checked with members from which the data had been solicited during the data analysis and in the final report to ensure that the interpretations had captured the meaning articulated in the interview (Korstjens & Moser, 2018).

Transferability

Even though qualitative studies such as the phenomenological approach tend to be unique in studying a particular phenomenon observed among a group of people, steps should be made to achieve a sense of generalizability (Guba & Lincoln, 1982). I first displayed transferability through participant selection. According to Guba and Lincoln (1982), purposive sampling is a good method to enhance the transferability of the study. With purposive sampling, the participants are chosen based on a specific purpose, providing rich and in-depth findings of the phenomenon being studied (Cypress, 2017). Additionally, a thick description of the methodology ensured transferability with an indepth explanation of the data collection, analysis, and final report (Cypress, 2017; Korstjens & Moser, 2018). A thick description will allow other researchers to determine

what aspects of the study can be transferable in a similar context (Guba & Lincoln, 1982).

Dependability

Data produced from phenomenological studies are from emerging themes, so exact replications cannot be achieved. However, the research must take steps to create stability in the data (Guba & Lincoln, 1982). To achieve dependability, I used an audit trail. In this process, I detailed all the research decisions and document all activities involved in the data collection, interview process, and data analysis (Korstjens & Moser, 2018). Koch (1994) argued that even if readers might develop different interpretations from the researcher, they need to know the steps and the decision process taken to arrive at the final report.

Confirmability

The researcher is also considered a vital instrument in qualitative research whose decision-making should be documented. Along with the same procedure used for dependability, I maintained a reflective diary to record the research process and explain the extent of my involvement with data analysis and interpretation. The information in the diary included thoughts, ideas, thought processes I went through when developing themes (Cypress, 2017; Guba & Lincoln, 1982; Korstjens & Moser, 2018).

Ethical Procedures

I only approached participants once I received IRB approval (#09-21-22-0420219). I placed flyers around African beauty salons, African churches, and international grocery stores (Appendix A). I informed women that their participation was

voluntary, and I did not use any coercion or incentive strategies. Participants had an informed consent form that explained the study and were aware that they could withdraw from the study at any point without any penalty. As a sign of gratitude for their time and willingness to share their experience, participants received a \$15 gift card.

To maintain confidentiality, I assigned a pseudonym to each participant. Only the signed consent form had the participant's signature, which I kept in a locked cabinet in my home office. Interviews were conducted in a location agreed upon with participants. All recorded materials were saved on my personal computer, protected by a password. I continued to use pseudonyms on the final report, presentations, and any following publications. All related study information, including paper and digital copies, will be kept for five years and deleted after that. Papers will be shredded, and electronic materials will be erased.

Summary

Qualitative interviews were used to provide information that was necessary to address Congolese immigrant women's satisfaction with care while receiving postpartum counseling in the U.S. Data coding required in-depth analysis and interpretation to tell participants' stories. The phenomenological approach was used to describe lived experiences. Chapter 4 includes results and explanations of the findings.

Chapter 4: Results

The purpose of the study was to examine lived experiences of Congolese immigrant women living in the U.S. with postpartum counseling and explore how counseling informs their decision-making processes. I also examined their interactions with healthcare services involving family planning procurement. To achieve this, I interviewed 10 women who described their experiences with family planning postpartum counseling, talked about their perceptions of their visits, and interactions with the healthcare system to obtain contraceptive methods. This chapter includes a discussion of the pilot study to explain how instrumentation was validated. I then present the setting, demographic characteristics of participants, data collection, and analysis. Findings are summarized at the end of the chapter. Quotations from participants were provided to support each theme.

Research Questions

RQ1: How do Congolese immigrants perceive their experiences with postpartum family planning counseling?

RQ2: How do postpartum family planning counseling experiences inform their decision-making processes?

RQ3: How do Congolese immigrants interact with healthcare systems for the procurement of modern family planning methods?

Pilot

The purpose of the pilot was to test the researcher-developed instrument to assess if interview questions would capture what they were designed to measure. After receiving

IRB approval (#09-21-22-0420219), I posted flyers around African business stores to recruit participants for the pilot study. Interested participants contacted me for a phone interview to confirm eligibility and schedule interviews. The pilot study included three participants. I followed the protocol outlined by the IRB by starting with the consent form, and continued after participants gave consent. Interviews took about 40 to 45 minutes, and I asked if questions were clear and understandable. I conducted all three meetings virtually because participants felt the option was more flexible, and they could schedule later in the evening. I summarized interviews and asked participants if I had articulated the meaning of their lived experiences. I also performed member-checking. The pilot study showed the instrument needed no changes.

Setting

After completing the pilot study, I posted flyers around African businesses (such as hair salons, grocery stores, and clothing stores) and churches with a sizeable Congolese congregation to recruit participants. Just as in the pilot study, I screened interested participants by phone to make sure they qualified for the study. Options for one-on-one interviews were either in person, virtually, or by phone. Most interviews were done virtually to accommodate fluctuating schedules. Only two people chose to be in person in a location of their choice. Interviews took 40 to 45 minutes. I conducted interviews until I felt that data had reached saturation. All participants lived in North Carolina. A gift card of \$15 was given to participants who volunteered for their time and efforts.

Demographics

All participants were female and between the ages of 26 and 45. The majority had associate degrees that were completed in the DRC. All participants were married and had at least two children, except one. Each participant had been in North Carolina for at least 6 years. Participants lived in the U.S. for 5 or more years and were familiar with the U.S. healthcare system. Eleven women were originally recruited, but one did not complete the interview, and her data were not used.

Table 1

Demographics

Participants	Pseudonyms	Education	Marital	Age	Number of
		Level	Status	Category	Children
Participant 1	Favor	Associate's	Married	26-35	4
Participant 2	Natasha	Associate's	Married	26-35	2
Participant 3	Alice	Associate's	Married	26-35	4
Participant 4	Rachel	Master's	Married	26-35	3
Participant 5	Pamela	Bachelor's	Married	26-35	2
Participant 6	Delilah	Associate's	Married	36-45	4
Participant 7	Ophelia	Associate's	Married	36-45	2
Participant 8	Nelly	HS Diploma	Married	26-35	3
Participant 9	Tia	HS Diploma	Married	26-35	1
Participant 10	Irene	Master's	Married	36-45	2

Data Collection

Final data were collected from 10 women. I assigned pseudonyms to protect identities. Qualitative data were collected through in-depth semi-structured one-on-one interviews. Most participants preferred virtual interviews through Zoom, and only two were done in person. Before proceeding with interviews, I reviewed the informed consent form, reiterating that participation was voluntary and information was confidential. For in-person interviews, I handed participants a consent form to sign. For virtual interviews,

I asked them to state the words, "I consent." Before starting interview questions, I asked participants demographic questions (see Appendix D). I then described the purpose of the research and interview process and allowed them a chance to ask any questions. I took some notes during interviews but not too much to avoid distracting participants. After interviews, I took more observation notes which served as a reflective diary. I began interviewing for the pilot and main study during December 2022 and finished in March 2023.

Data Analysis

I used manual transcription because a third party might struggle with accents and phrases. Interviews were transcribed verbatim via Microsoft Word by listening and reviewing audio multiple times to ensure I captured all information and answers. I conducted member checking by sending participants a summary of their interviews to verify whether I had understood their experiences via email. I then imported all transcripts to NVivo v.12 for analysis. Thematic analysis was used. I built themes from multiple codes. This approach allowed me to identify, analyze, and interpret patterns of meaning as lived by participants (Maguire & Delahunt, 2017).

I first had to familiarize myself with data, including demographic documents, interview notes, transcripts from audio recordings, and field notes. I later used them for codes and bracketing. As I was reviewing, I highlighted statements that I believed were essential to participants' experiences and stood out. I added notes to margins of transcriptions for later reference in my analysis. During the second phase, I started to generate initial codes. I again highlighted any words and statements that were interesting.

The third phase consisted of searching for themes. This was achieved by finding repeating statements and observations of similar contexts. I then reduced statements to categories and eventually developed brackets of understanding. During the fourth phase, I reviewed potential themes. I started eliminating, moving, or changing my mind about sorting concepts. The fifth phase was defining themes, explaining them, and ensuring they reflected my research questions. The final phase consisted of writing a report to explain the lived phenomenon from data with quotes to support each theme.

Based on the analysis, five themes arose from the data: complexity of the U.S. healthcare system, familiarity with healthcare providers, immigrant status, perceived understanding of family planning, and factors associated with procurement.

Evidence of Trustworthiness

Credibility

Credibility allows the researcher to remain accurate in the data collection and analysis, safeguarding against subjectiveness in the study. I ensured credibility through member checking to promote collaboration between myself and the participants (Korstjens & Moser, 2018). I made a summary of the interview and checked with each participant to make sure that I understood their perspectives. Since English is not their native language, meanings can be misinterpreted while explaining experiences and telling a story. All analysis was done after I received confirmation from participants that their words were well summarized.

Transferability

Even though generalizability is not expected in qualitative research, various methods exist to demonstrate applicability of the study. The process allows other researchers to evaluate concepts of this study that can be applicable for other situations. I demonstrated transferability by keeping a thick description to retain a detailed account of my experience during data collection, analysis, and final report (Cypress, 2017; Korstjens & Moser, 2018).

Dependability

I used journal notes during and especially after the interviews to document participants' interesting statements or gestures. I noted any observations during interviews and questions I wanted to follow up on from a participant's answer. Also, I used an audit trail to document my process of identifying unique codes and theme development. I maintained a log of all the coding in NVivo (Korstjens & Moser, 2018).

Confirmability

I used a reflective diary to ensure that the experiences described in the study reflect the views of the participants and not mine as a researcher. I took notes of any biases and insights I gained after an interview; I asked participants to clarify any statements that were not clear to me. Keeping a reflective diary allowed me to put my biases aside so that they do not influence the findings and interpretations (Cypress, 2017; Guba & Lincoln, 1982; Korstjens & Moser, 2018).

Results

The interviews began by asking participants about their overall experience with giving birth in the United States. Participants shared their experience with birth delivery in a new country as I felt it was necessary to understand their experience with postpartum care and as an initiation to the overall U.S. healthcare system. I then discussed their experience with postpartum family planning counseling and their general perception. I concluded the interviews by asking participants about their interaction with the healthcare system for the procurement of contraceptive methods. I used meaningful statements as in vivo coding, and they became the baseline to develop themes.

RQ1

This question was intended to gain an understanding of the experience of Congolese women during postpartum family planning counseling. Three themes emerged in answering this question using interview questions 9, 10, 11, and 12 (Table 2).

Table 2

Emerging Themes for RQ1

Research Questions	Interview Questions	Emergent Themes
RD1. How do Congolese immigrants perceive their experiences with postpartum family planning counseling?	1. Can you tell me about yourself and your experience with giving birth here in the United States?	Theme 1: Complexity of the US healthcare system Subtheme 1: The U.S. Healthcare System is Highly Organized
	2. Can you describe to me an instance when you went to a postpartum family planning counseling visit?	Subtheme 2: Obligation to See a Healthcare Provider Subtheme 3: Packed Information in a Short Amount of Time

3. What was your experience with your healthcare provider during postpartum family planning counseling?

Theme 2: Familiarity with Healthcare Providers

4. How did your status as an African immigrant contribute to the service you experienced?

4. How did your status **Theme 3: Immigrant Status**

Theme 1: Complexity of the Healthcare System

Theme 1 yielded three subthemes as women talked about their experience with giving birth in general, then explained some surprising elements associated with receiving counseling within the healthcare system.

Subtheme 1: The U.S. Healthcare System is Highly Organized. Participants were asked about their overall experience with giving birth in the United States to set the stage for understanding their encounter with reproductive health services. All ten participants agreed that their stay in the hospital was a positive experience when compared to how things are done in their home country, DRC. For example, words like "comfortable," "safe," "private rooms," and "organize" were often mentioned. Natasha elaborated on her first experience giving birth in the United States. She talked about all the prenatal visits and the health portal that allowed her to check everything from her phone. Pamela commented on how safe she felt being at the hospital for delivery. She

added that nurses were "caring and protective of my health and the baby's, especially when visitors were there."

Subtheme 2: Obligation to See a Healthcare Provider. As participants began to describe an instance when they went for a postpartum family planning visit, all women noted the need to see a doctor whenever they needed to decide on reproductive care.

Natasha stated "I did not go to my first postnatal at six weeks because I was so overwhelmed. I was still in discomfort from my c-section and did not want to go outside. But then, when I was interested in birth control, I realized I must go to the hospital. Well, this is the reality here: you must see a doctor for everything. For the second baby, I went to postpartum 6-week visits to talk about my contraception needs. I felt it would help reduce the number of times I need to see a doctor."

Delilah mentioned that choosing a contraceptive method can also dictate how often you need to see a doctor. Some contraception methods, such as the IUD, can be provider-dependent. Alice said "The problem with the IUD is that you must see the doctor every time there is an issue. If you think about having it removed, well...it's another trip to the hospital".

Subtheme 3: Packed Information in a Short Amount of Time. Another subtheme that emerged as all participants talked about their experience was the overwhelming amount of information given in a short time. In this subtheme, women reported feeling that everything in the United States always felt like a hurry, with doctors and nurses packing a lot of information quickly. Like the hospital stay after a baby's delivery, postpartum family planning counseling visits are short despite important details

needing to be discussed. Tia said "It was just like when I gave birth at the hospital.

Nurses kept coming into the room with much information and paper for you to sign.

Sometimes, you don't even remember the instructions they give you...When I scheduled a contraceptive visit, it was only for 30 minutes, and you must discuss everything in less than 30 minutes because you also have to do some hospital administration stuff."

At postpartum counseling sessions, women reported experiencing the same dump of information as during delivery. Rachel added "The nurse had to go through so many methods in a short amount of time that all she kept giving me was their percent effectiveness. The first time I went to my contraception counseling, I thought it would be more like a conversation. I did not want it to be like a list of dos and don'ts. It was all technical."

Theme 2: Familiarity with Healthcare Providers

Participants were asked about their overall experience with their healthcare providers during a postpartum family planning counseling session. Findings on this theme were mixed, with four participants remembering a positive experience with their healthcare providers. Alice said "I became more comfortable with my provider by the second baby. I felt comfortable talking about my feelings and emotions and felt like I could go back to her with any questions".

Three participants reported a negative experience with their healthcare providers, with some feeling like there was no closeness with the provider, so it was challenging to open up too much about any concerns. Nelly added "When I had my first baby, I never had the same Ob/gyn during my antenatal visits, at delivery, and when I went for

postpartum family planning counseling. The process did not allow me to become familiar with a provider and voice all my concerns".

For one participant, the experience was plain "horrible," feeling the healthcare provider's disrespectful and condescending attitude. Pamela said "I told my provider that I wanted to stop using the patch because it was bothering me, making me feel angry all the time. But she did not believe me. Then she was like, oh well, whatever you want". Pamela summarized that an exchange like this makes the conversation less enjoyable and leaves one wanting the counseling session to end.

Theme 3: Immigrant Status

When participants were asked if their status as an African immigrant woman contributed to their service experience, interviewees thought their ability to speak English and communicate their needs were paramount. All participants felt their status as African immigrants did not affect their experience and answered "no" to this question. However, three women expressed how fluency in English could impact how you receive counseling. Tia, who mentioned having used some interpreters when she first arrived in the United States during hospital visits, said "I'm glad I do not need interpretation services because some providers do not like to go through them."

RQ2
Table 3
Emerging Themes for RQ2

Research Questions	Interview Questions	Emergent Themes
RQ2. How do postpartum	5. How did the	Theme 4: Perceived
family planning counseling	information	Understanding of
experiences inform their	received during	Family planning

decision-making processes?	postpartum family planning counseling influence your	Subtheme 1: Perceived Side Effects and
	decision to use contraception?	Subtheme 2: Importance of Birth Spacing
6	. Before receiving postpartum family planning counseling in the United States, did you know about family planning methods before?	Subtheme 3: Previous Knowledge as an Empowering Tool

Theme 4: Perceived Understanding of Family Planning

There were three subthemes that best summarized how family planning counseling sessions informed women's decision-making.

Subtheme 1: Information on Side Effects. Participants were asked how the information from the postpartum family planning counseling sessions helped inform their decisions. Irene stated "I wanted to make sure that the method I chose will help me easily resume trying to conceive. My friend had the injection and said that when she wanted to get pregnant again, she couldn't, and she was never able to get pregnant again. That's why I went with the patch."

Women felt like side effects were not explained correctly. Discussing side effects was often overlooked, and information focused mainly on the method's effectiveness. Tia said "they said that the IUD makes you stop bleeding every month after some time. I did not feel comfortable with that. I am not sure if it is good for your body to stop bleeding." Alice added "With the IUD, my bleeding was slightly unpredictable." Favor also

confirmed "I had the IUD after the third baby, but I had to ask them to take it out because I was feeling unstable. My mood changed too much". Pamela said, "I had the patch, and often I was easily irritable, always dealing with mood swings."

Subtheme 2: Importance of Birth Spacing. Along with the importance of the information received from postpartum family planning counseling sessions, birth spacing was cited as an essential factor for choosing contraceptive methods for nine women.

Favor explained "I already had a method in my mind because I had a timeframe of when I want to get pregnant again. I just wanted to make sure that I choose a method that I can discontinue quickly and resume fertility". Alice said, "I wanted to have IUD put in after my 2nd child because I decided to go back to school to receive an associate degree".

Only one woman, Ophelia, mentioned that birth spacing was not important for her because here goal was to limit births. She said: "I did not have counseling with my first baby because I was ready to have another one. But I chose a permanent method for my second because I wanted to focus on my career".

Aside from pursuing a career, three women wanted to talk about birth spacing as a way to recover and be healthy for another pregnancy. Nelly argued: "Many of my friends told me that the long method made it difficult to get pregnant again. I just wanted contraception because I had a c-section and wanted to make sure I didn't get pregnant immediately, you know, to allow my wound to heal".

Subtheme 3: Previous Knowledge as an Empowering Tool. This subtheme emerged from asking participants if they knew about family planning methods before receiving postpartum counseling in the United States. Six participants had received some

education about contraceptives in the past. They mentioned that the forms of contraception discussed in the visits were not so foreign to them. This knowledge was, for some, a motivator to go for counseling and contraceptive uptake. Natasha said "A group of educators used to come to the university to talk about contraception. Since I was young and not planning to get married soon, I did not ask many questions. However, I was aware of most contraceptive methods". Rachel added "At our church, we had ladies' meetings where they used to address contraception to space births. Because maternal mortality is so high back in the country, we were encouraged to discuss contraception with our physician after getting married".

RQ3

One theme emerged regarding Congolese immigrant women's experiences involving procuring family planning methods (see Table 4).

Table 4

Emerging Themes for RQ3

Research Questions	Interview Questions	Emergent Themes
RQ3. How do Congolese immigrants interact with healthcare systems for the procurement of modern family planning methods?	7. Describe your experience with the healthcare system in getting any form of contraception.	Theme 5: Factors associated with Procurement Subtheme 1: Convenience of Some Family Planning Methods Procurement
	8. Did this experience affect the type of contraception you decided to use?9. How, if at all, did your type of insurance contribute to your choice of	Subtheme 2: (Affordability) Burden of Unknown Healthcare Costs

Theme 5: Factors Associated with Procurement

Two subthemes surfaced when participants talked about their interaction with the healthcare system, mainly pharmacies, to obtain family planning methods. They highlighted the convenience of procurement and the burden of unknown healthcare costs to afford contraceptives.

Subtheme 1: Convenience. Convenience In this subtheme, participants described their experience with the healthcare system for procurement of family planning methods and how this interaction impacted their overall decision. Seven women who chose short-term contraceptive methods felt that going to the pharmacy was convenient. Rachel said "I don't like to go to the doctor's office all the time, you know. For example, when I return to work, I must ask for time off, and I don't get paid for that day. So, I got the pills. My friend had injections and had to return to a doctor's office often. I liked the fact that you could go to the closest pharmacy, go to the drive-through window, and get it without needing to get out of the car. Whenever I need a refill, I could stop by the pharmacy after work."

Subtheme 2: Affordability. In the final part of the survey, participants were asked if the type of insurance contributed to their choice of contraceptive methods. All participants mentioned discussing cost-sharing with their healthcare providers during counseling sessions to ensure their insurance plan type had little or no out-of-pocket expense. Rachel shared this "When I had my first baby, they told me there would be a

copay with my insurance if I went with the IUD. I did not want that. Here in the United States, you always get a surprise bill after a visit to the hospital. I had to think about minimizing my expenses. But I did not care about the cost after my second baby because I was not interested in having another baby". Another woman, Delilah, expressed frustration with medical bills and did not want to deal with the out-of-pocket cost of contraception.

Summary

Five themes emerged from research questions to understand lived experiences of Congolese women going through postpartum family planning counseling and discover how counseling informs their decision-making processes. Furthermore, themes also helped to understand interaction experiences with the healthcare system involving family planning procurement. A total of 10 participants were recruited and able to share their experiences with postpartum family planning counseling and interactions with the healthcare system involving contraception procurement. Each participant answered nine open-ended interview questions following the same data collection process. Data collection and analysis followed to ensure integrity of participants' views and minimize researcher bias.

What developed from interviews is that participants felt the importance of birth spacing during their reproductive journey, which drove their decision-making processes during postpartum family planning counseling sessions. Another theme that emerged among all participants was that the U.S. healthcare system was highly organized, and most policy requirements were in place to maximize maternal and infant positive health

outcomes. The significance of themes is further discussed in Chapter 5, where interpretations, limitations, further recommendations, and implications of the study are addressed.

Chapter 5: Discussion, Conclusions, and Recommendations

This chapter includes interpretations of findings from Chapter 4. The purpose of this study was to examine lived experiences of Congolese immigrant women living in the U.S. with postpartum counseling and explore how counseling informs their decision-making processes as well as interactions with healthcare services for family planning procurement. To achieve these aims, a phenomenological approach was employed using semi-structured one-on-one qualitative interviews.

Use of postpartum family planning counseling differs among immigrant women, with this particular group being less likely to receive high-quality counseling on postpartum contraception. Consequently, there are disparities in terms of contraception uptake and maternal outcomes compared to U.S.-born women (Coleman-Minahan & Potter, 2019). Previous research has predominantly focused on Latina women, examining their attitudes, beliefs, perceptions, and experiences with family planning and counseling. However, as the number of African immigrant women in the U.S. is increasing, studies have begun to explore their family planning use patterns and experiences with postpartum services, including counseling. Most studies have primarily concentrated on refugee populations from countries such as the DRC, Somalia, or Ethiopia. Additionally, none of these studies investigated how African immigrant women navigate the healthcare system to access family planning methods. This study contributes to existing literature by addressing experiences of Congolese immigrant women from the DRC. Through interviews, five themes emerged: (a) complexity of the U.S. healthcare system, (b) familiarity with healthcare providers, (c) immigrant status, (d) perceived understanding of family planning, (e) factors associated with procurement. This chapter not only includes interpretations of these findings but also discusses study limitations, recommendations, implications for social change, and a conclusion.

Interpretation of the Findings

The experience of participants during postpartum family planning counseling and navigating the healthcare system for procurement is discussed and compared to previous studies in Chapter 2. Given the increasing number of Congolese immigrant women and importance of promoting postpartum contraceptive use, it is crucial to comprehend the experiences of Congolese women during family planning counseling.

Theme 1: Complexity of the U.S. Healthcare System

Participants in the study highlighted challenges when interacting with the healthcare system for maternal services and postpartum family planning counseling. They acknowledged that for first-time users, the system can be overwhelming and difficult to navigate. Language barriers and insurance issues are common complexities within the healthcare system (Banke-Thomas et al., 2019; Bustamante et al., 2019; McKenney et al., 2018). Participants in this study added the quality aspect of service as another complexity, expressing concerns about rushed services provided by healthcare providers.

Interviews revealed another dimension of the U.S. healthcare system, which is the necessity of consulting a healthcare provider to obtain contraceptives. All participants emphasized that the postpartum period is already challenging due to responsibilities of caring for a newborn and fulfilling pediatrician visit obligations. However, any decision related to family planning and its uptake necessitates a consultation with a healthcare

provider. This discovery aligns with previous research on both U.S.-born women and immigrant women. For example, participants raised concerns about drawbacks of long-term contraceptive methods such as IUDs. These include the requirement to consult a doctor for the insertion and removal of the IUD, as well as any potential complications that may arise (Janiak et al., 2018; Namasivayam et al., 2022).

Participants provided further insights regarding the brevity of family planning counseling sessions, comparing them to hospital stays during childbirth. They expressed a sense of rushed visit times, leaving them with limited opportunities to ask questions or voice concerns. Because of tightly-scheduled systems, interactions between healthcare providers and patients are constrained, focusing primarily on discussing one or two family planning methods while overlooking other concerns. This issue of limited counseling duration has been identified by both U.S.-born and immigrant women (Manzer & Bell, 2022; Thiel de Bocanegra et al., 2020). It is crucial to acknowledge that nonnative English speakers often require additional time to effectively communicate their needs.

Theme 2: Familiarity with Healthcare Providers

All participants felt having the same healthcare provider for other gynecological and obstetric care, especially one who was personally chosen, made the experience with counseling positive and comfortable. Although coercion was not mentioned, some participants reported experiencing unsupportive attitudes from providers when expressing a preference for a specific contraceptive method. This negative experience often occurred when participants had to see a healthcare provider whom they were not familiar with. For

Nelly and Pamela, receiving maternal care and postpartum counseling from different healthcare providers hindered smooth patient-provider interactions. At the time, they were informed that their Medicaid coverage required them to see different providers.

By contrast, participant Alice expressed her satisfaction with her OBGYN doctor, whom she had for all gynecologic, maternal, and reproductive health care for over 4 years. Alice felt she could ask her doctors anything and send online messages via the hospital patient's online portal knowing she could trust their doctor's answers. As a result, Alice reported a positive overall experience during postpartum family planning counseling sessions. In a study by Ahad et al. (2019), African immigrant women who had consistent primary healthcare providers experienced a 15.0% increase in health literacy compared to those without a regular doctor. Although Ahad's research concentrated on primary care, shared experiences of Congolese women from this current study emphasizes the crucial role of consistent healthcare providers in empowering women to make informed decisions regarding family planning methods.

Latina women expressed a desire to consult with multiple healthcare providers, as they perceived certain providers exhibited bias towards specific methods and certain racial and ethnic communities. However, they emphasized that initiating family planning counseling during the prenatal period is the optimal time to engage with multiple providers (Sznajder et al., 2020).

Theme 3: Immigrant Status

Participants did not perceive their immigrant status as influencing their experience. Somali immigrant women were more focused on ways they were treated in

terms of their Islamic religion, which was part of their culture and beliefs, rather than being immigrants (Agbemenu, Volpe et al., 2018).

Fluency in the English language had an impact on experiences of certain participants. Regarding use of interpreters, two participants expressed their dissatisfaction, finding it unpleasant. Kolak et al. (2022) indicated immigrant women felt uncomfortable discussing contraception due to its private nature and believed interpreters were not conveying everything accurately. Gele et al. (2020) claimed some Somali-born women desired the presence of professional interpreters. This study's findings contribute to existing research by demonstrating that counseling experiences involving postpartum family planning may not necessarily differ based on immigration status, but rather culture and comfort with the English language.

Theme 4: Perceived Understanding of Family Planning

Side effects were extensively discussed during individual interviews. Participants acknowledged that certain side effects, such as mood swings or irregular periods, were anticipated. Nevertheless, they expressed their dissatisfaction with healthcare providers who often failed to provide comprehensive information about these side effects and how one might feel about them, particularly due to appointment time constraints. Additionally, some participants mentioned instances where healthcare providers disregarded their concerns or displayed a dismissive attitude when providers believed the side effects were unfounded.

Participants reported a positive experience with postpartum family planning counseling and contraceptive decision when the conversation was centered around birth

spacing rather than limiting births. Women recognized the significance of family planning methods due to reasons such as emergency c-sections (which required some time to recover from the surgery before another pregnancy). In this case, they preferred counseling that focused on short-acting methods rather than long-acting methods like IUDs if they wanted a relatively shorter interbirth intervals. When women had the desire to pursue education or employment opportunities, they were fine with IUDs. This finding aligns with previous studies that highlight how Somali women's perceptions on the importance of family planning methods for birth spacing changed after immigrating to the United States as they became aware of reality of the new host country (Olorunsaiye et al., 2023; Robinet et al., 2023; Zhang et al., 2020).

Participants with a higher level of education, like Natasha and Rachel, demonstrated prior knowledge and awareness of family planning methods. They believed that this knowledge empowered them during discussions with their healthcare provider and reduced feelings of coercion (Berndt & Bell, 2021). This observation is consistent with another study that revealed Congolese women exhibited greater familiarity with family planning methods and a higher level of acceptance compared to Somali women (Royer et al., 2020).

Theme 5: Factors Associated with Procurement

Overall, the participants expressed satisfaction with the process of obtaining family planning methods, while highlighting concerns regarding affordability. They found that the methods they desired were readily available, particularly for short-acting options like the pill. The presence of drive-through pharmacies and extended operating

hours were convenient and helped overcome certain barriers to accessing family planning methods. However, for long-acting contraceptives, acquiring them required additional visits to the doctor's office. Nevertheless, once inserted, these methods did not require any ongoing maintenance from the user.

The affordability of family planning methods posed a concern for women, particularly those with private insurance and Medicaid for Pregnant Women (also called Emergency Medicaid). Women with private insurance expressed worry about copayments and unexpected expenses, as they already felt burdened by the costs associated with hospital baby deliveries. However, when it came to choosing long-acting or permanent family planning methods after completing their desired number of children, Rachel, one of the participants, did not express any concerns regarding the expenses involved. She stated that the out-of-pocket cost for a permanent family planning method is more reasonable compared to the expenses associated with raising another child. Additionally, participants who were covered by traditional Medicaid, which continues to cover medical expenses beyond pregnancies, did not have any worries about the financial aspect. These findings align with previous studies conducted on U.S.-born and immigrant Latina women, which also highlighted the advantage of traditional Medicaid coverage when choosing any types of family planning methods (Rodriguez et al., 2022; Rodriguez, Skye, et al., 2021; Yates et al., 2022).

The inclusion of discussions on the convenience (accessibility) and affordability of family planning methods in this topic contributes to the existing literature, providing insights into the reasons behind the frequent use of short-acting as opposed to long-acting

contraceptives among African women immigrants (Agbemenu et al., 2020; Cope et al., 2019; Rodriguez, Kaufman, et al., 2021).

Interpretation of the Findings and SCT

The understanding of the results obtained in this research study is demonstrated through the application of the Social Cognitive Theory (SCT) and the incorporation of the constructs of self-efficacy and outcome expectancy. Bandura's SCT explains how personal, behavioral, and environmental factors interact to motivate individuals to adopt new health behaviors (Chess et al., 1978; Wood & Bandura, 1989). In order to support women in making informed choices regarding family planning, postpartum family planning counseling serves as an intervention that provides information to facilitate decision-making and encourages the initiation of contraceptive use. Hence, the effectiveness of counseling is crucial in ensuring satisfaction and empowering women to continue using contraception as desired (Coleman-Minahan & Potter, 2019).

Bandura's constructs of self-efficacy and outcome expectancy within the SCT suggests that individuals are more inclined to engage in a particular behavior when they possess both the belief in their ability to effectively perform the necessary actions and the expectation of achieving a desired outcome (Bandura, 2004; Chess et al., 1978). In the context of my research, all participants had made a contraceptive decision and adopted its use. The research findings highlighted factors that impacted the participants' self-efficacy and expected outcome motivators while engaging in postpartum family planning counseling and navigating the healthcare system for method procurement. These factors could either strengthen or undermine their self-efficacy. Information about these factors

can be utilized to educate healthcare providers on these factors, enabling them to enhance self-efficacy and gain a better understanding of the expectations of Congolese women when making contraceptive decisions during family planning counseling.

Self-Efficacy

Proficiency in the English language was found to enhance comfort during patientprovider interactions. Hence, it played a role in strengthening women's self-efficacy in making informed decisions, as they were able to comprehend the information provided by the provider.

The discussion regarding family planning methods and their potential side effects was deemed important by the participants, as it positively influenced their self-efficacy. Participants expressed a desire for healthcare providers to exhibit patience and attentiveness when addressing concerns about side effects, as it could impact the acceptability and adherence to the chosen method. Additionally, participants who possessed prior knowledge of contraception reported a positive experience with counseling and were able to engage in discussions about methods they were already familiar with.

Finally, the participants' self-efficacy was influenced by the procurement of family planning methods. The type of insurance they had significantly influenced their experience with the healthcare system. When women had access to their preferred method, it increased their satisfaction and confidence in using the method. Conversely, when women's contraceptive options were limited due to cost, it impacted their contraceptive decision-making.

The findings indicated that short-duration family planning counseling sessions, and unfamiliarity with healthcare providers undermined self-efficacy. Participants expressed that rushed appointments resulted in limited time for establishing trust and discussing personal concerns. Moreover, unfamiliarity with healthcare providers hindered meaningful patient-provider interactions, potentially leading to cold-toned family planning counseling.

Outcome Expectancy

The participants shared a positive encounter, and a key motivator that influenced their decision-making was when the counseling emphasized the significance of spacing pregnancies. This was especially important for reasons such as pursuing educational or employment opportunities and recovering from a medically difficult childbirth like emergency c-section.

Limitations of the Study

Despite the rigorous application of qualitative research principles, the study had a few limitations. One such limitation was that participants needed to be comfortable conducting a 40–45-minute interview in English, which resulted in the exclusion of individuals who lacked confidence in conversing in English. This restriction may have hindered the exploration of different perspectives on women with limited English proficiency and their ability to share their stories. However, it is worth noting that the study's findings align with previous research on African immigrants in the United States, where participants were able to freely express themselves in their native language with the help of interpreters.

Another limitation is that interpretation of the findings in this study may not accurately represent the experiences of recently arrived Congolese women, as the participants had already resided in the United States for a minimum of five years. However, despite this limitation, the study's findings will still make a valuable contribution to the existing research on postpartum family planning and service utilization among African immigrant communities in the United States.

Recommendations

Conducting research plays a crucial role in improving the healthcare system and guiding clinical practice. Evidence-based research offers valuable data and findings that support healthcare providers and public health workers in developing effective interventions. With the rise in immigration and permanent resettlement of non-U.S. born women, it is imperative to establish public health policies and practices that can improve the quality of care for diverse immigrant populations. Although there is a growing body of research on family planning counseling and behaviors among African immigrant women, it is important to recognize the cultural and belief-based variations among them. Therefore, further studies are needed to explore the experiences of African women in the United States from different countries (Okeke-Ihejirika et al., 2018; Olorunsaiye et al., 2023).

The findings of this study provide valuable recommendations for future research and practice. The recommendation for future research is to focus on exploring the experiences of healthcare providers when interacting with African immigrant women, beyond the existing research on Somali women (Agbemenu, Volpe, et al., 2018; Banke-

Thomas et al., 2019; Egeh et al., 2019; Zhang et al., 2020). This research should aim to understand the challenges healthcare providers face and identify the similarities and differences in delivering family planning counseling to various African immigrant women. By utilizing the findings from these studies, policies can be developed to enhance patient satisfaction and provide cultural competency training to healthcare professionals, enabling effective communication and foster trust.

The first recommendation for practice is to increase the utilization of telehealth medicine for family planning counseling. The COVID-19 pandemic and its aftermath have led to the emergence of digital health, which has transformed the way healthcare providers can reach out to underserved communities (Keesara et al., 2020). This option is particularly beneficial for English-speaking African immigrant women as it reduces the need for frequent visits to healthcare facilities, promotes more patient-provider interactions, facilitates decision-making, and enhances the overall experience of postpartum family planning counseling. A study conducted by Hill et al. (2021) found no significant difference in the use of telemedicine versus in-clinic services among Latina women. Women who were unable to utilize telemedicine mentioned reasons such as the lack of Spanish services or translation availability (Hill et al., 2021).

The second recommendation for practice is for healthcare providers to exercise patience and empathy when discussing family planning options with patients, taking into consideration the various factors that influence their decisions. It is crucial for providers to recognize that immigrant women may prioritize their cultural values and beliefs while

also grappling with the challenges of adapting to a new country and its realities (Omenka et al., 2020).

Implications

The findings provide valuable insights for healthcare providers and policymakers in comprehending the lived experience of Congolese women in postpartum family planning counseling, as well as the factors that influence their decision-making.

Additionally, it enables them to identify factors that affect their access to family planning methods in a complex healthcare system. The experiences shared by Congolese women in this study shed light on certain aspects that can enhance the delivery of services. This is of utmost importance as it aids healthcare providers and policymakers in promoting the adoption of family planning methods across diverse racial and ethnic groups to reduce disparities, while empowering women to sustain family planning uptake.

Public health goes beyond clinical settings to enhance patient outcomes by exploring community-based approaches, such as culturally appropriate education. Participants who reported having prior knowledge of family planning methods received this information as part of community programs (e.g. public health workers going to universities). Consequently, the findings indicate that community-level programs that emphasize teaching the advantages and disadvantages of different family planning methods can empower Congolese women with knowledge to increase self-efficacy and reduce the pressure to make hasty decisions during brief family planning counseling (Dolan et al., 2020; Gele et al., 2020).

Conclusions

Through this study, Congolese immigrant women have contributed their personal experiences with counseling and navigating the healthcare system for family planning methods. This study is the first to explore these aspects within a non-refugee population of Congolese immigrant women in the United States. Through a phenomenological study involving 10 participants, five themes emerged that can enhance the overall experience of immigrant women during family planning counseling. These themes include the complexity of the U.S. healthcare system, familiarity with healthcare providers, immigrant status, perceived understanding of family planning, and factors related to procurement. The participants understood the importance of confidentiality and believed that sharing their experiences would contribute valuable data to research and daily practice. This study's findings are innovative and essential to public health, as they provide unique insights from a population that has been underrepresented in previous research.

References

- Agbemenu, K., Auerbach, S., Ely, G. E., Aduloji-Ajijola, N., & Wang, H. (2020).

 Family planning trends among community-dwelling African refugee women.

 Public Health Nursing, 37(4), 478–486. https://doi.org/10.1111/phn.12725
- Agbemenu, K., Auerbach, S., Murshid, N. S., Shelton, J., & Amutah-Onukagha, N. (2019). Reproductive health outcomes in African refugee women: A comparative study. *Journal of Women's Health*, 28(6), 785–793. https://doi.org/10.1089/jwh.2018.7314
- Agbemenu, K., Hannan, M., Kitutu, J., Terry, M. A., & Doswell, W. (2018). "Sex will make your fingers grow thin and then you die": The interplay of culture, myths, and taboos on African immigrant mothers' perceptions of reproductive health education with their daughters aged 10-14 years. *Journal of Immigrant and Minority Health*, 20(3), 697–704.

 https://doi.org/10.1007/s10903-017-0675-4
- Agbemenu, K., Mencia, J. J., de Rosa, C., Aidoo-Frimpong, G., & Ely, G. (2022).

 Family planning research in African Immigrant and refugee women: A scoping review. *Journal of Transcultural Nursing*.

 https://doi.org/10.1177/10436596211072891
- Agbemenu, K., Volpe, E. M., & Dyer, E. (2018). Reproductive health decision-making among US-dwelling Somali Bantu refugee women: A qualitative study. *Journal of Clinical Nursing*. https://doi.org/10.1111/jocn.14162

- Ahad, F. B., Zick, C. D., Simonsen, S. E., Mukundente, V., Davis, F. A., & Digre,
 K. (2019). Assessing the likelihood of having a regular health care provider among African America n and African Immigrant Women. *Ethnicity and Disease*, 29(2), 253–260. https://doi.org/10.18865/ed.29.2.253
- Ahrens, K. A., Thoma, M. E., Copen, C. E., Frederiksen, B. N., Decker, E. J., & Moskosky, S. (2018). Unintended pregnancy and interpregnancy interval by maternal age, National Survey of Family Growth. *Contraception*, 98(1), 52–55. https://doi.org/10.1016/j.contraception.2018.02.013
- Åkerman, E., Larsson, E. C., Essén, B., & Westerling, R. (2019). A missed opportunity? Lack of knowledge about sexual and reproductive health services among immigrant women in Sweden. *Sexual and Reproductive Healthcare*, 19, 64–70. https://doi.org/10.1016/j.srhc.2018.12.005
- Anderson, M. (2017). African immigrant population in U.S. steadily climbs.

 https://policycommons.net/artifacts/618796/african-immigrant-population-in-us/1599797/
- Aspers, P., & Corte, U. (2019). What is qualitative in qualitative research.

 *Qualitative Sociology, 42(2), 139–160. https://doi.org/10.1007/s11133-019-9413-7
- Aztlan-James, E. A., McLemore, M., & Taylor, D. (2017). Multiple unintended pregnancies in U.S. Women: A systematic review. *Women's Health Issues*, 27(4), 407–413. https://doi.org/10.1016/j.whi.2017.02.002

- Bandura, A. (1986). Social foundations of thought and action: A social cognitive theory. Prentice-Hall, Inc.
- Bandura, A. (2004). Health promotion by social cognitive means. *Health Education* and *Behavior*, 31(2), 143–164. https://doi.org/10.1177/1090198104263660
- Banke-Thomas, A., Agbemenu, K., & Johnson-Agbakwu, C. (2019). Factors associated with access to maternal and reproductive health care among Somali refugee women resettled in Ohio, United States: A cross-sectional survey. *Journal of Immigrant and Minority Health*, 21(5), 946–953. https://doi.org/10.1007/s10903-018-0824-4
- Barrett, D., & Twycross, A. (2018). Data collection in qualitative research.

 Evidence-Based Nursing 21(3), 63–64. https://doi.org/10.1136/eb-2018-102939
- Bearak, J., Popinchalk, A., Alkema, L., & Sedgh, G. (2018). Global, regional, and subregional trends in unintended pregnancy and its outcomes from 1990 to 2014: Estimates from a Bayesian hierarchical model. *Lancet Global Health*, 6(4), e380–e389. https://doi.org/10.1016/S2214-109X(18)30029-9
- Bergen, N., & Labonté, R. (2020). "Everything is perfect, and we have no problems:

 Detecting and limiting social desirability bias in qualitative research.

 Qualitative Health Research, 30(5), 783–792.

 https://doi.org/10.1177/1049732319889354
- Bernard, H. R. (2017). Research methods in anthropology: Qualitative and quantitative approaches. Rowman & Littlefield.

- Berndt, V. K., & Bell, A. V. (2021). "This is what the truth is": Provider-patient interactions serving as barriers to contraception. *Health (United Kingdom)*, 25(5), 613–629. https://doi.org/10.1177/1363459320969775
- Berry, J. W. (2005). Acculturation: Living successfully in two cultures.

 *International Journal of Intercultural Relations, 29(6 SPEC. ISS.), 697–712.

 https://doi.org/10.1016/j.ijintrel.2005.07.013
- Boydell, V., Neema, S., Wright, K., & Hardee, K. (2018). Closing the gap between people and programs: Lessons from implementation of social accountability for family planning and reproductive health in Uganda. *African Journal of Reproductive Health*, 22(1), 73–84.

 https://doi.org/10.29063/ajrh2018/v22i1.7
- Braun, V., & Clarke, V. (2021). To saturate or not to saturate? Questioning data saturation as a useful concept for thematic analysis and sample-size rationales. *Qualitative Research in Sport, Exercise and Health*, *13*(2), 201–216. https://doi.org/10.1080/2159676X.2019.1704846
- Brunson, M. R., Klein, D. A., Olsen, C. H., Weir, L. F., & Roberts, T. A. (2017).

 Postpartum contraception: initiation and effectiveness in a large universal healthcare system. *American Journal of Obstetrics and Gynecology*, 217(1), 55.e1-55.e9. https://doi.org/10.1016/j.ajog.2017.02.036
- Bustamante, A. V., Chen, J., McKenna, R. M., & Ortega, A. N. (2019). Health Care

 Access and Utilization Among U.S. Immigrants Before and After the

- Affordable Care Act. *Journal of Immigrant and Minority Health*, 21(2), 211–218. https://doi.org/10.1007/s10903-018-0741-6
- Carvajal, D. N., Gioia, D., Mudafort, E. R., Brown, P. B., & Barnet, B. (2017). How can Primary Care Physicians Best Support Contraceptive Decision Making?

 A Qualitative Study Exploring the Perspectives of Baltimore Latinas.

 Women's Health Issues, 27(2), 158–166.

 https://doi.org/10.1016/j.whi.2016.09.015
- Carvajal, D. N., Mudafort, P. C. R., Barnet, B., & Blank, A. E. (2020).

 Contraceptive Decision Making Among Latina Immigrants: Developing
 Theory-Based Survey Items. *Hispanic Health Care International*, *18*(4),
 181–190. https://doi.org/10.1177/1540415319883422
- Chand, M. (2019). Brain Drain, Brain Circulation, and the African Diaspora in the United States. *Journal of African Business*, 20(1), 6–19. https://doi.org/10.1080/15228916.2018.1440461
- Chess, S., Hassibi, M., Chess, S., & Hassibi, M. (1978). Theories of Child

 Development. *Principles and Practice of Child Psychiatry*, 6, 45–72.

 https://doi.org/10.1007/978-1-4899-2198-7_3
- Clark, K. R., & Vealé, B. L. (2018). Strategies to enhance data collection and analysis in qualitative research. *Radiologic Technology*, 89(5), 482CT-485CT.

- Coleman-Minahan, K., & Potter, J. E. (2019). Quality of postpartum contraceptive counseling and changes in contraceptive method preferences. *Contraception*, 100(6), 492–497. https://doi.org/10.1016/j.contraception.2019.08.011
- Cooper, M., & Cameron, S. (2018). Postpartum contraception. *Obstetrics, Gynaecology and Reproductive Medicine*, 28(6), 183–185.

 https://doi.org/10.1016/j.ogrm.2018.04.003
- Cope, A. G., Mara, K. C., Weaver, A. L., & Casey, P. M. (2019). Postpartum

 Contraception Usage among Somali Women in Olmsted County, Minnesota.

 Obstetrics and Gynecology, 133(4), 762–769.

 https://doi.org/10.1097/AOG.0000000000003154
- Covington-Ward, Y., Agbemenu, K., & Matambanadzo, A. (2018). "We feel like it was better back home:" Stress, Coping, and Health in a U.S. Dwelling

 African Immigrant Community. *Journal of Health Care for the Poor and Underserved*, 29(1), 253–265. https://doi.org/10.1353/hpu.2018.0018
- Creswell, J. W. (2014). The Selection of a Research Approach. Research design:

 Qualitative, quantitative, and mixed methods approaches. Thousand Oaks,

 CA: Sage Publications
- Cross-Barnet, C., Courtot, B., Hill, I., Benatar, S., Cheeks, M., & Markell, J. (2018).

 Facilitators and Barriers to Healthy Pregnancy Spacing among Medicaid

 Beneficiaries: Findings from the National Strong Start Initiative. *Women's*Health Issues, 28(2), 152–157. https://doi.org/10.1016/j.whi.2017.12.004

- Cypress, B. S. (2017). Rigor or reliability and validity in qualitative research:

 Perspectives, strategies, reconceptualization, and recommendations.

 Dimensions of Critical Care Nursing, 36(4), 253–263.

 https://doi.org/10.1097/DCC.0000000000000253
- Dehlendorf, C., Anderson, N., Vittinghoff, E., Grumbach, K., Levy, K., & Steinauer, J. (2017). Quality and Content of Patient–Provider Communication About Contraception: Differences by Race/Ethnicity and Socioeconomic Status.

 Women's Health Issues, 27(5), 530–538.*

 https://doi.org/10.1016/j.whi.2017.04.005
- Dehlendorf, C., Reed, R., Fox, E., Seidman, D., Hall, C., & Steinauer, J. (2018).

 Ensuring our research reflects our values: The role of family planning research in advancing reproductive autonomy. In *Contraception* (Vol. 98, Issue 1, pp. 4–7). Elsevier USA.

 https://doi.org/10.1016/j.contraception.2018.03.015
- DeSisto, C. L., Hirai, A. H., Collins, J. W., & Rankin, K. M. (2018). Deconstructing a disparity: explaining excess preterm birth among U.S.-born black women.

 Annals of Epidemiology, 28(4), 225–230.

 https://doi.org/10.1016/J.ANNEPIDEM.2018.01.012
- Díez, E., Ló Pez, M. J., Marí-Dell'olmo, M., Nebot, L., Pé Rez, G., Villalbi, J. R., & Carreras, R. (2017). Effects of a counselling intervention to improve contraception in deprived neighbourhoods: a randomized controlled trial.

- The European Journal of Public Health, 28(1), 10–15. https://doi.org/10.1093/eurpub/ckx046
- Dolan, H., Li, M., Bateson, D., Thompson, R., Tam, C. W. M., Bonner, C., & Trevena, L. (2020). Healthcare providers' perceptions of the challenges and opportunities to engage Chinese migrant women in contraceptive counselling: A qualitative interview study. *Sexual Health*, *17*(5), 405–413. https://doi.org/10.1071/SH19215
- Downey, M. M., Arteaga, S., Villaseñor, E., & Gomez, A. M. (2017). More Than a Destination: Contraceptive Decision Making as a Journey. *Women's Health Issues*, 27(5), 539–545. https://doi.org/10.1016/j.whi.2017.03.004
- Echeverria-Estrada, C., & Batalova, J. (2019). Sub-Saharan African Immigrants in the United States. Migration Policy Institute .

 https://www.migrationpolicy.org/article/sub-saharan-african-immigrants-united-states-2018
- Egeh, A. A., Dugsieh, O., Erlandsson, K., & Osman, F. (2019). The views of Somali religious leaders on birth spacing A qualitative study. *Sexual and Reproductive Healthcare*, *20*(February), 27–31. https://doi.org/10.1016/j.srhc.2019.02.003
- Ejike, C., Lartey, G., Capps, R., & Ciochetty, D. (2020). Health-seeking cultural patterns in the use of available healthcare services among refugee groups in south-central Kentucky. *International Journal of Migration, Health and*

- Social Care, 16(1), 12–21. https://doi.org/10.1108/IJMHSC-08-2018-0051/FULL/XML
- Evans, N. M., Spates, K., Kirkland, C., & Mubikayi Kabasele, C. (2022). Voices of Resettled Refugee Congolese Women: A Qualitative Exploration of Challenges Associated with Resettling in Ohio. *Journal of Refugee Studies*, 35(1), 1–15. https://doi.org/10.1093/JRS/FEAB068
- Ewerling, F., Victora, C. G., Raj, A., Coll, C. V. N., Hellwig, F., & Barros, A. J. D. (2018). Demand for family planning satisfied with modern methods among sexually active women in low- and middle-income countries: Who is lagging behind? *Reproductive Health*, *15*(1). https://doi.org/10.1186/s12978-018-0483-x
- Fertman, C. I., & Allensworth, D. D. (Eds.) (2016). Theory in Health Promotion

 Programs. *Health promotion programs: From theory to practice* (pp. 53-77).

 John Wiley & Sons.
- Fox, M., Thayer, Z., Wadhwa, P. D., & Thayer, Z. M. (2017). Acculturation and health: the moderating role of socio-cultural context HHS Public Access. *Am Anthropol*, *119*(3), 405–421. https://doi.org/10.1111/aman.12867
- Frechette, J., Bitzas, V., Aubry, M., Kilpatrick, K., & Lavoie-Tremblay, M. (2020).

 Capturing Lived Experience: Methodological Considerations for Interpretive

 Phenomenological Inquiry. *International Journal of Qualitative Methods*, 19.

 https://doi.org/10.1177/1609406920907254

- Gele, A. A., Musse, F. K., & Qureshi, S. (2019). Unmet needs for contraception: A comparative study among Somali immigrant women in Oslo and their original population in Mogadishu, Somalia. *PLOS ONE*, *14*(8), e0220783. https://doi.org/10.1371/JOURNAL.PONE.0220783
- Gele, A. A., Musse, F. K., Shrestha, M., & Qureshi, S. (2020). Barriers and facilitators to contraceptive use among Somali immigrant women in Oslo: A qualitative study. *PLOS ONE*, *15*(3). https://doi.org/10.1371/journal.pone.0229916
- Glanz, K., Rimer, B. K., & Viswanath, K. (Eds.). (2015). How Individuals,

 Environments, And Health Behaviors Interact Social Cognitive Theory.

 Health behavior: Theory, research, and practice (5th ed.) (pp.169-188). San

 Francisco, CA: Jossey-Bass. 169-188
- Glasier, A., Bhattacharya, S., Evers, H., Gemzell-Danielsson, K., Hardman, S., Heikinheimo, O., la Vecchia, C., & Somigliana, E. (2019). Contraception after pregnancy. *Acta Obstetricia Gynecologica Scandinavica*, *98(11)*, 1378–1385. https://doi.org/10.1111/aogs.13627
- Guba, E. G. (1981). Criteria for assessing the trustworthiness of naturalistic inquiries. *ECTJ*, 29(2), 75. https://doi.org/10.1007/BF02766777
- Guba, E. G., & Lincoln, Y. S. (1982). Epistemological and methodological bases of naturalistic inquiry.
- Hanley, G. E., Hutcheon, J. A., Kinniburgh, B. A., & Lee, L. (2017). Interpregnancy interval and adverse pregnancy outcomes an analysis of successive

- pregnancies. *Obstetrics and Gynecology*, *129*(3), 408–415. https://doi.org/10.1097/AOG.0000000000001891
- Hernandez, N. D., Chandler, R., Nava, N., Tamler, I., Daley, E. M., Baldwin, J. A., Buhi, E. R., O'Rourke, K., Romero-Daza, N., & Grilo, S. (2020). Young adult US-born Latina women's thoughts, feelings and beliefs about unintended pregnancy. *Culture, Health and Sexuality*, 22(8), 920–936. https://doi.org/10.1080/13691058.2019.1642517
- Hill, B. J., Lock, L., & Anderson, B. (2021). Racial and ethnic differences in family planning telehealth use during the onset of the COVID-19 response in Arkansas, Kansas, Missouri, and Oklahoma. *Contraception*, 104(3), 262–264. https://doi.org/10.1016/j.contraception.2021.05.016
- Ho, L. S., & Wheeler, E. (2018). Using Program Data to Improve Access to Family Planning and Enhance the Method Mix in Conflict-Affected Areas of the Democratic Republic of the Congo. *Global Health: Science and Practice*, 161–177. www.ghspjournal.org
- Im, H., & Swan, L. E. T. (2019). Qualitative exploration of critical health literacy among Afghan and Congolese refugees resettled in the USA. *Health Education Journal*, 78(1), 38–50.
 https://doi.org/10.1177/0017896918785932
- Ismail, N., Kinchin, G., & Edwards, J.-A. (2017). Pilot Study, Does It Really

 Matter? Learning Lessons from Conducting a Pilot Study for a Qualitative

- PhD Thesis. *International Journal of Social Science Research*, *6*(1), 1. https://doi.org/10.5296/ijssr.v6i1.11720
- Jackson, A. v., Wang, L. F., & Morse, J. (2017). Racial and ethnic differences in contraception use and obstetric outcomes: A review. In *Seminars in Perinatology* (Vol. 41, Issue 5, pp. 273–277). W.B. Saunders. https://doi.org/10.1053/j.semperi.2017.04.003
- Janiak, E., Clark, J., Bartz, D., Langer, A., & Gottlieb, B. (2018). Barriers and Pathways to Providing Long-Acting Reversible Contraceptives in Massachusetts Community Health Centers: A Qualitative Exploration.

 *Perspectives on Sexual and Reproductive Health, 50(3), 111–118.

 https://doi.org/10.1363/psrh.12071
- Johnson, J. L., Adkins, D., & Chauvin, S. (2020). A review of the quality indicators of rigor in qualitative research. In *American Journal of Pharmaceutical Education* (Vol. 84, Issue 1, pp. 138–146). American Association of Colleges of Pharmacy. https://doi.org/10.5688/ajpe7120
- Jones, R. K. (2017). Change and consistency in US women's pregnancy attitudes and associations with contraceptive use. *Contraception*, *95*(5), 485–490. https://doi.org/10.1016/j.contraception.2017.01.009
- Kadarisman, M. (2019). Family Planning Program in the National Population and Family Planning Board.

- Kantorova, V. (2020). Unintended pregnancy and abortion: what does it tell us about reproductive health and autonomy? *The Lancet*, 8. https://doi.org/10.1016/S2214109X(20)303429
- Kavanaugh, M. L., & Jerman, J. (2018). Contraceptive method use in the United States: trends and characteristics between 2008, 2012 and 2014.

 **Contraception, 97(1), 14–21.

 https://doi.org/10.1016/j.contraception.2017.10.003
- Keesara, S., Jonas, A., & Schulman, K. (2020). Covid-19 and Health Care's Digital Revolution. *New England Journal of Medicine*, *382*(23), e82. https://doi.org/10.1056/nejmp2005835
- Kim, T. Y., Dagher, R. K., & Chen, J. (2016). Racial/Ethnic Differences in Unintended Pregnancy: Evidence from a National Sample of U.S. Women. American Journal of Preventive Medicine, 50(4), 427–435. https://doi.org/10.1016/j.amepre.2015.09.027
- Koch, R. (1994). is the professional code still the cornerstone of clinical practice. In JAN Forum The Author. Journal compilation \acute{O} (Vol. 19, Issue 1). Blackwell Publishing Ltd.
- Kolak, M., Löfgren, C., Hansson, S. R., Rubertsson, C., & Agardh, A. (2022).

 Immigrant women's perspectives on contraceptive counselling provided by midwives in Sweden–a qualitative study. *Sexual and Reproductive Health Matters*, 30(1). https://doi.org/10.1080/26410397.2022.2111796

- Korstjens, I., & Moser, A. (2018). Series: Practical guidance to qualitative research.

 Part 4: Trustworthiness and publishing. *European Journal of General*Practice, 24(1), 120–124. https://doi.org/10.1080/13814788.2017.1375092
- López, G., & Bialik, K. (2017). Key findings about {U}.{S}. immigrants. *Pew Research Center*, 1960, 1–15.
- Lopez, L. M., Grey, T. W., Chen, M., Tolley, E. E., & Stockton, L. L. (2016).

 Theory-based interventions for contraception. In *Cochrane Database of Systematic Reviews* (Vol. 2016, Issue 11). John Wiley and Sons Ltd.

 https://doi.org/10.1002/14651858.CD007249.pub5
- Lynn Woodgate, R., Shiyokha Busolo, D., Crockett, M., Dean, R. A., Amaladas, M. R., & Plourde, P. J. (2017). A qualitative study on African immigrant and refugee families' experiences of accessing primary health care services in Manitoba, Canada: it's not easy! *International Journal for Equity in Health*, 16(5), 1–13. https://doi.org/10.1186/s12939-016-0510-x
- Maguire, M., & Delahunt, B. (2017). *Doing a Thematic Analysis: A Practical, Step-by-Step Guide for Learning and Teaching Scholars.* * (Issue 3). http://ojs.aishe.org/index.php/aishe-j/article/view/335
- Manzer, J. L., & Bell, A. v. (2022). "Did I Choose a Birth Control Method Yet?":

 Health Care and Women's Contraceptive Decision-Making. *Qualitative*Health Research, 32(1), 80–94. https://doi.org/10.1177/10497323211004081
- McKenney, K. M., Martinez, N. G., & Yee, L. M. (2018). Patient navigation across the spectrum of women's health care in the United States. In *American*

- Journal of Obstetrics and Gynecology (Vol. 218, Issue 3, pp. 280–286). Mosby Inc. https://doi.org/10.1016/j.ajog.2017.08.009
- McMorrow, S., & Saksena, J. (2017). Voices and Views of Congolese Refugee

 Women: A Qualitative Exploration to Inform Health Promotion and Reduce

 Inequities. *Health Education and Behavior*, 44(5), 769–780.

 https://doi.org/10.1177/1090198117726572
- Mehta, P. K., Mody, K. S. D., & Raj, S. S. C. A. (2018). Learning from UJAMBO:

 Perspectives on Gynecologic Care in African Immigrant and Refugee

 Women in Boston, Massachusetts. *Journal of Immigrant and Minority*Health, 20(2), 380–387. https://doi.org/10.1007/s10903-017-0659-4
- Metusela, C., Ussher, J., Perz, J., Hawkey, A., Morrow, M., Narchal, R., Estoesta, J., & Monteiro, M. (2017). "In My Culture, We Don't Know Anything About That": Sexual and Reproductive Health of Migrant and Refugee Women. International Journal of Behavioral Medicine, 24(6), 836–845. https://doi.org/10.1007/s12529-017-9662-3
- Millar, A., Vogel, R. I., Bedell, S., Ayers Looby, M., Hubbs, J. L., Harlow, B. L., & Ghebre, R. (2017). Patterns of postpartum contraceptive use among Somali immigrant women living in Minnesota. *Contraception and Reproductive Medicine*, 2(1), 14. https://doi.org/10.1186/s40834-017-0041-x
- Moniz, M., Chang, T., Heisler, M., & Dalton, V. K. (2017). Immediate postpartum long-acting reversible contraception: the time is now. *Contraception*, *95*(4), 335–338. https://doi.org/10.1016/j.contraception.2016.11.007

- Moser, A., & Korstjens, I. (2018). Series: Practical guidance to qualitative research.

 Part 3: Sampling, data collection and analysis. In *European Journal of General Practice* (Vol. 24, Issue 1, pp. 9–18). Taylor and Francis Ltd.

 https://doi.org/10.1080/13814788.2017.1375091
- Muruthi, B., Watkins, K., McCoy, M., Muruthi, J. R., & Kiprono, F. J. (2017). "I Feel Happy that I Can be Useful to Others": Preliminary Study of East African Women and Their Remittance Behavior. *Journal of Family and Economic Issues*, 38(3), 315–326. https://doi.org/10.1007/s10834-017-9533-8
- Naderifar, M., Goli, H., & Ghaljaie, F. (2017). Snowball Sampling: A Purposeful Method of Sampling in Qualitative Research. *Strides in Development of Medical Education*, *14*(3). https://doi.org/10.5812/sdme.67670
- Namasivayam, A., Schluter, P. J., Namutamba, S., & Lovell, S. (2022).

 Understanding the contextual and cultural influences on women's modern contraceptive use in East Uganda: A qualitative study. *PLOS Global Public Health*, *2*(8), e0000545. https://doi.org/10.1371/journal.pgph.0000545
- Nawyn, S. J., & Park, J. (2019). Gendered segmented assimilation: earnings trajectories of African immigrant women and men. *Ethnic and Racial Studies*, 42(2), 216–234. https://doi.org/10.1080/01419870.2017.1400085
- Ngendahimana, D., Amalraj, J., Wilkinson, B., Verbus, E., Montague, M., Morris, J., & Arora, K. S. (2021). Association of race and ethnicity with postpartum

- contraceptive method choice, receipt, and subsequent pregnancy. *BMC Women's Health*, 21(1). https://doi.org/10.1186/s12905-020-01162-8
- Nyoni, S., Chihoho, T., & Nyoni, T. (2021). Projection of Total Fertility Rate (TFR) in the Democratic Republic of Congo. *International Research Journal of Innovations in Engineering and Technology*, 5(8), 115–118. https://www.proquest.com/docview/2605657396?pq-origsite=gscholar&fromopenview=true
- Oakley, L. P., Harvey, S. M., & López-Cevallos, D. F. (2018). Racial and Ethnic Discrimination, Medical Mistrust, and Satisfaction with Birth Control Services among Young Adult Latinas. *Women's Health Issues*, 28(4), 313–320. https://doi.org/10.1016/j.whi.2018.03.007
- Okeke-Ihejirika, P., Salami, B., & Karimi, A. (2018). African immigrant women's experience in western host societies: a scoping review. *Journal of Gender Studies*, 27(4), 428–444. https://doi.org/10.1080/09589236.2016.1229175
- Oliphant, S. M. (2019). Voices of Ethiopian immigrant women. *International Social Work*, 62(2), 581–594. https://doi.org/10.1177/0020872817742700
- Oliver, E. A., Klebanoff, M., Yossef-Salameh, L., Oza-Frank, R., Moosavinasab, S., Reagan, P., Muglia, L., Buhimschi, C. S., & Buhimschi, I. A. (2018).

 Preterm birth and gestational length in four race-nativity groups, including Somali Americans. *Obstetrics and Gynecology*, *131*(2), 281–289.

 https://doi.org/10.1097/AOG.0000000000000002427

- Olorunsaiye, C. Z., Brunner Huber, L. R., & Ouedraogo, S. P. (2023). Interbirth

 Intervals of Immigrant and Refugee Women in the United States: A CrossSectional Study. *International Journal of Maternal and Child Health and*AIDS (IJMA), 12(1). https://doi.org/10.21106/ijma.621
- Olukotun, O., & Mkandawire-Valhmu, L. (2020). Lessons Learned From the

 Recruitment of Undocumented African Immigrant Women for a Qualitative

 Study. *International Journal of Qualitative Methods*, 19.

 https://doi.org/10.1177/1609406920904575
- Omani-Samani, R., Ranjbaran, M., Mohammadi, M., Esmailzadeh, A., Sepidarkish, M., Maroufizadeh, S., & Almasi-Hashiani, A. (2019). Impact of Unintended Pregnancy on Maternal and Neonatal Outcomes. *Journal of Obstetrics and Gynecology of India*, 69(2), 136–141. https://doi.org/10.1007/s13224-018-1125-5
- Omenka, O. I., Watson, D. P., & Hendrie, H. C. (2020). Understanding the healthcare experiences and needs of African immigrants in the United States:

 A scoping review. In *BMC Public Health* (Vol. 20, Issue 1). BioMed Central Ltd. https://doi.org/10.1186/s12889-019-8127-9
- Opoku-Dapaah, E. (2017). "Ain't Making It in America": The Economic

 Characteristics of African Immigrants in North Carolina, USA. *Journal of Immigrant and Refugee Studies*, 15(4), 406–427.

 https://doi.org/10.1080/15562948.2016.1234091

- Pakseresht, S., Rasekh, P., & Leili, E. K. (2018). Physical health and maternal-fetal attachment among women: Planned versus unplanned pregnancy.

 *International Journal of Women's Health and Reproduction Sciences, 6(3), 335–341. https://doi.org/10.15296/ijwhr.2018.55
- Pazol, K., Robbins, C. L., Black, L. I., Ahrens, K. A., Daniels, K., Chandra, A., Vahratian, A., & Gavin, L. E. (2017). Receipt of Selected Preventive Health Services for Women and Men of Reproductive Age United States, 2011–2013. MMWR. Surveillance Summaries, 66(20), 1–31. https://doi.org/10.15585/mmwr.ss6620a1
- Pinter, B., Hakim, M., Seidman, D., Kubba, A., Kishen, M., & di Carlo, C. (2017).

 Religion and family planning. TT -. European Journal of Contraception and Reproductive Health Care, 21(6), 486–495.

 https://doi.org/http://dx.doi.org/10.1080/13625187.2017.1295438
- Potter, J. E., Hubert, C., & White, K. (2017). The Availability and Use of

 Postpartum LARC in Mexico and Among Hispanics in the United States.

 Maternal and Child Health Journal, 21(9), 1744–1752.

 https://doi.org/10.1007/s10995-016-2179-6
- Quinn, D. A., Sileanu, F. E., Zhao, X., Mor, M. K., Judge-Golden, C., Callegari, L. S., & Borrero, S. (2020). History of unintended pregnancy and patterns of contraceptive use among racial and ethnic minority women veterans.

 *American Journal of Obstetrics and Gynecology, 223(4), 564.e1-564.e13. https://doi.org/10.1016/j.ajog.2020.02.042

- Qutoshi, S. B. (2018). Journal of Education and Educational Development

 Discussion Phenomenology: A Philosophy and Method of Inquiry (Vol. 5,

 Issue 1).
- Ray, M. K., & King, V. (2018). Immediate Pospartum LARC: An Underused Contraceptive Option. *American Family Physician*, 97(1), 9–11.
- Refugee Processing Center. (2017). Admissions and Arrivals: FY 2017 Arrivals by State and Nationality. https://www.wrapsnet.org/archives/
- Refugee Processing Center. (2022). Admissions and Arrivals: FY 2022 Arrivals by State and Nationality. https://www.wrapsnet.org/archives/.
- Robinet, L., Jeffredo, A., & Clesse, C. (2023). Factors Influencing Contraceptive

 Choice During the Postpartum Period: A Qualitative Systematic Review.

 Journal of Midwifery & Women's Health.

 https://doi.org/10.1111/jmwh.13471
- Rodriguez, M. I., Kaufman, M., Lindner, S., Caughey, A. B., Defede, A. L., & McConnell, K. J. (2021). Association of Expanded Prenatal Care Coverage for Immigrant Women with Postpartum Contraception and Short Interpregnancy Interval Births. *JAMA Network Open*.
 https://doi.org/10.1001/jamanetworkopen.2021.18912
- Rodriguez, M. I., McConnell, K. J., Skye, M., Kaufman, M., Caughey, A. B., Lopez-Defede, A., & Darney, B. G. (2022). Disparities in postpartum contraceptive use among immigrant women with restricted Medicaid

- benefits. AJOG Global Reports, 2(1). https://doi.org/10.1016/j.xagr.2021.100030
- Rodriguez, M. I., Skye, M., Lindner, S., Caughey, A. B., Lopez-Defede, A., Darney,
 B. G., & McConnell, K. J. (2021). Analysis of Contraceptive Use among
 Immigrant Women following Expansion of Medicaid Coverage for
 Postpartum Care. *JAMA Network Open*, 4(12).
 https://doi.org/10.1001/jamanetworkopen.2021.38983
- Romano, M., Cacciatore, A., Giordano, R., & La Rosa, B. (2010). Postpartum period: three distinct but continuous phases. *Journal of Prenatal Medicine*, 4(2), 22–25.
- Rosenthal, L., & Lobel, M. (2020). Gendered racism and the sexual and reproductive health of Black and Latina Women. *Ethnicity and Health*, 25(3), 367–392. https://doi.org/10.1080/13557858.2018.1439896
- Royer, P. A., Olson, L. M., Jackson, B., Weber, L. S., Gawron, L., Sanders, J. N., & Turok, D. K. (2020). "In Africa, There Was No Family Planning. Every Year You Just Give Birth": Family Planning Knowledge, Attitudes, and Practices Among Somali and Congolese Refugee Women After Resettlement to the United States. *Qualitative Health Research*, 30(3), 391–408. https://doi.org/10.1177/1049732319861381
- Saksena, J., & McMorrow, S. (2017). We Really Do Have the Same Goals: The Push and Pull of One Community–Academic Partnership to Support

- Congolese Refugee Women. *Narrative Inquiry in Bioethics*, 7(1), 9–12. https://doi.org/10.1353/nib.2017.0004
- Saksena, J., & McMorrow, S. L. (2019). Through their Eyes: A Photovoice and Interview Exploration of Integration Experiences of Congolese Refugee

 Women in Indianapolis. *Journal of International Migration and Integration*2019 21:2, 21(2), 529–549. https://doi.org/10.1007/S12134-019-00672-1
- Sawhill, I. V., & Guyot, K. (2019). Preventing unplanned pregnancy: lessons from the states. *Economic Studies at Brookings*.
- Schivone, G. B., & Glish, L. L. (2017). Contraceptive counseling for continuation and satisfaction. *Current Opinion in Obstetrics and Gynecology*, 29(6), 443–448. https://doi.org/10.1097/GCO.0000000000000000000
- Semu, L. L. (2020). The Intersectionality of Race and Trajectories of AfricanWomen into the Nursing Career in the United States. *Behavioral Sciences*, 10(4). https://doi.org/10.3390/bs10040069
- Showers, F. (2018). Learning to care: Work experiences and identity formation among african immigrant care workers in the US. *International Journal of Care and Caring*, 2(1), 7–25. https://doi.org/10.1332/239788218X15187914933434
- Spierling, T., & Shreffler, K. M. (2018). Tough Decisions: Exploring Women's Decisions Following Unintended Pregnancies. Frontiers in Sociology, 3. https://doi.org/10.3389/fsoc.2018.00011

- Sridhar, A., & Salcedo, J. (2017). Optimizing maternal and neonatal outcomes with postpartum contraception: impact on breastfeeding and birth spacing.

 *Maternal Health, Neonatology, and Perinatology, 3(1), 1–10.

 https://doi.org/10.1186/s40748-016-0040-y
- Stephenson, J., Bailey, J. v, Gubijev, A., D'souza, P., Oliver, S., Blandford, A., Hunter, R., Shawe, J., Rait, G., Brima, N., Copas, A., & Garrett, E. (2020).

 An interactive website for informed contraception choice: randomised evaluation of Contraception Choices. *Digital Health*, 6, 1–16.

 https://doi.org/10.1177/2055207620936435
- Sundstrom, B., Szabo, C., & Dempsey, A. (2018). "My Body. My Choice": A

 Qualitative Study of the Influence of Trust and Locus of Control on

 Postpartum Contraceptive Choice. *Journal of Health Communication*, 23(2),

 162–169. https://doi.org/10.1080/10810730.2017.1421728
- Sznajder, K., Carvajal, D. N., & Sufrin, C. (2020). Patient perceptions of immediate postpartum long-acting reversible contraception: A qualitative study,.

 Contraception, 101(1), 21–25.

 https://doi.org/10.1016/j.contraception.2019.09.007
- Tamir, C. (2022). *Key findings about Black immigrants in the U.S.* | *Pew Research Center*. Pew Research Center. https://www.pewresearch.org/fact-tank/2022/01/27/key-findings-about-black-immigrants-in-the-u-s/
- Tapales, A., Douglas-Hall, A., & Whitehead, H. (2018). The sexual and reproductive health of foreign-born women in the United States.

- Contraception, 98(1), 47–51. https://doi.org/10.1016/j.contraception.2018.02.003
- Thiel de Bocanegra, H., Kenny, J., Sayler, K., Turocy, M., & Ladella, S. (2020).

 Experiences with Prenatal and Postpartum Contraceptive Services among

 Women with a Preterm Birth. *Women's Health Issues*, *30*(3), 184–190.

 https://doi.org/10.1016/j.whi.2019.11.002
- Tran, N. T., Gaffield, M. E., Seuc, A., Landoulsi, S., Yamaego, W. M. E., Cuzin-Kihl, A., Kouanda, S., Thieba, B., Mashinda, D., Yodi, R., Kiarie, J., & Reier, S. (2018). Effectiveness of a package of postpartum family planning interventions on the uptake of contraceptive methods until twelve months postpartum in Burkina Faso and the Democratic Republic of Congo: The YAM DAABO study protocol. *BMC Health Services Research*, *18*(1). https://doi.org/10.1186/S12913-018-3199-2
- Tran, N. T., Seuc, A., Tshikaya, B., Mutuale, M., Landoulsi, S., Kini, B., Mbu
 Nkolomonyi, B., Nyandwe Kyloka, J., Langwana, F., Cuzin-Kihl, A., Kiarie,
 J., Gaffield, M. E., Yodi, R., & Mashinda Kulimba, D. (2020). Effectiveness of post-partum family planning interventions on contraceptive use and method mix at 1 year after childbirth in Kinshasa, DR Congo (Yam Daabo):
 a single-blind, cluster-randomised controlled trial. *The Lancet Global Health*, 8(3), e399–e410. https://doi.org/10.1016/S2214-109X(19)30546-7
- Tran, N. T., Yameogo, W. M. M., Langwana, F., Kouanda, S., Thieba, B.,

 Mashinda, D., Yodi, R., Nyandwe Kyloka, J., Millogo, T., Coulibaly, A.,

- Zan, S., Kini, B., Ouedraogo, B., Puludisi, F., Cuzin-kihl, A., Reier, S., Kiarie, J., & Gaffield, M. E. (2018). Birth spacing and informed family planning choices after childbirth in Burkina Faso and the Democratic Republic of Congo: Participatory action research to design and evaluate a decision-making tool for providers and their clients. *Patient Education and Counseling*, *101*(10), 1871–1875. https://doi.org/10.1016/j.pec.2018.05.004
- Troutman, M., Rafique, S., & Plowden, T. C. (2020). Are higher unintended pregnancy rates among minorities a result of disparate access to contraception? *Contraception and Reproductive Medicine*, *5*(1). https://doi.org/10.1186/s40834-020-00118-5
- Tucker, C., Berrien, K., Menard, M. K., Herring, A. H., Rowley, D., & Halpern, C.
 T. (2020). Preterm Birth and Receipt of Postpartum Contraception Among
 Women with Medicaid in North Carolina. *Maternal and Child Health* Journal, 24(5), 640–650. https://doi.org/10.1007/s10995-020-02889-4
- United Nations High Commissioner for Refugees. (2018). *UNHCR global*resettlement statistical report. Retrieved

 from http://www.unhcr.org/52693bd09.html
- U.S. Department of Homeland Security (2018). Yearbook of Immigration Statistics 2018. Retrieved from https://www.dhs.gov/immigrationstatistics/yearbook/2018

- U.S. Department of State & Bureau of Population, Refugees, and Migration.(2019). Refugee admission by region. Arlington, VA: Worldwide RefugeeAdmissions Processing System, Refugee Processing Center
- Williams, H. R., Goad, L., Treloar, M., Ryken, K., Mejia, R., Zimmerman, M. B., Stockdale, C., & Hardy-Fairbanks, A. (2019). Confidence and readiness to discuss, plan and implement postpartum contraception during prenatal care versus after delivery. *Https://Doi.Org/10.1080/01443615.2019.1586853*, 39(7), 941–947. https://doi.org/10.1080/01443615.2019.1586853
- Wood, R., & Bandura, A. (1989). Social Cognitive Theory of Organizational

 Management. *The Academy of Management Review*, *14*(3), 361.

 https://doi.org/10.2307/258173
- World Health Organization. (2010). Measuring sexual health: Conceptual and practical considerations and related indicators.

 www.who.int/reproductivehealth
- World Health Organization. (2017). Sexual and Reproductive Health.
- World Health Organization. (2018). Family Planning: A Global Handbook for Providers.
- Yates, L., Birken, S., Thompson, T. A., Stuart, G. S., Greene, S., Hassmiller Lich, K., & Weinberger, M. (2022). A qualitative analysis of Medicaid beneficiaries perceptions of prenatal and immediate postpartum contraception counseling. *Women's Health*, 18.
 https://doi.org/10.1177/17455057221124079

- Zhang, Y., McCoy, E. E., Scego, R., Phillips, W., & Godfrey, E. (2020). A

 Qualitative Exploration of Somali Refugee Women's Experiences with

 Family Planning in the U.S. *Journal of Immigrant and Minority Health*,

 22(1), 66–73. https://doi.org/10.1007/s10903-019-00887-5
- Zivich, P. N., Kawende, B., Lapika, B., Behets, F., & Yotebieng, M. (2019). Effect of Family Planning Counseling After Delivery on Contraceptive Use at 24 Weeks Postpartum in Kinshasa, Democratic Republic of Congo. *Maternal and Child Health Journal*, 23(4), 530–537. https://doi.org/10.1007/s10995-018-2667-y

Appendix A: Recruitment Flyer



Appendix B: Informed Consent

Informed Consent for Dissertation Research Participation: The Lived Experience of

I understand that to participate in this study, I will be asked questions such as education level, marital status, number of children, employment status, insurance type, and current family planning use.

I understand that Mrs. Nshisso will only select individuals who she believes will fit best into the following criteria for participation in the study:

- 1. Are legal immigrants in the United States and at least 18 years old (will have a valid state-issued identification card)
- 2. Have migrated to the United States within the last five years
- 3. Have had a child born in the United States within the last five years
- 4. Have received postpartum family planning counseling at the hospital following a birth or at a follow-up postpartum visit
- 5. Have sought and tried to navigate the healthcare system for contraception procurement
- 6. Can speak English

I understand that if I am not initially selected for participation in the study, my information will be kept until the completion of the study should the need for additional participants arise. If more participants are requested, I will be contacted to assess my interest in participating in the study.

I understand that my participation in this study will involve answering questions provided to me as a hard copy of my experience. Participation involves being interviewed by Lemba Nshisso and will last approximately 30-45 minutes. Notes will be written during the interview. An audiotape of the interview and subsequent dialogue will be made. If I don't want to be taped, I will not participate in the study. Tapes will be transcribed for analysis and destroyed when the study is complete.

I understand that most interviewees will find the discussion engaging and thoughtprovoking. If, however, I feel uncomfortable during the interview session, I have the right to decline to answer any question or end the interview. I understand that this study aims to contribute to the body of research by understanding how healthcare providers can best interact with Congolese immigrant women and discuss family planning methods during counseling.

I understand that the researcher will not identify me by name in any reports using information obtained from this interview. My confidentiality as a participant in this study will remain secure.

I understand that Mrs. Nshisso will give me an executive summary of my transcribed interview after her dissertation. I have the right to withdraw from this study at any time.

At the end of this study, I understand that all documentation that is matched with my name will be kept in a secured location along with all written documentation, computer files, audiotapes, identifying information, and all other information related to me. The reason for this is to ensure confidentiality. All materials will be destroyed after five years. Under this condition, I agree that any information obtained from this research may be used in any way thought best for publication or education, if I am in no way identified and my name is not used.

I understand that the study will be shared with Ms. Nshisso's dissertation committee at Walden University. Disguised extracts from my interview may be quoted in her dissertation or publications. I agree that any information obtained from this research may be used in any way thought best for publication or education if I remain anonymous.

I understand a gift card for \$15 will be provided as a gratuity for my participation in this study. Whether I complete the research study or decide to withdraw, I am still entitled to the gift card.

I understand that minimal personal risk and no physical discomfort are directly involved with this research.

I understand that if I have any questions or problems that arise in connection with my participation in this study, I should contact Mrs. Nshisso, the researcher, at 704.787.1717 (Email at lemba.nshisso@waldenu.edu). For research problems or questions regarding subjects, the Institutional Review Board may be contacted through [information of the contact person at Walden IRB office]. Walden University's approval number for this study is [insert approval number], and it expires [insert date].

I have read and understood the explanation provided to me. I have had all my questions answered to my satisfaction, and I voluntarily agree to participate in this study. I have been given a copy of this consent form.

Date:	Signature of Participant
Date:	Signature of Investigator

Appendix C: Demographic Information and Interview Questions

Demographic Information

1. Education Level			
Less than High School	☐ High School	Associate Degree	
University Degree	Advanced Degree	ee	
2. Marital Status			
Single	☐ Married	☐ Separated	
Divorced	Living with Part	ner	
☐ Wish not to disclose			
3. Age category			
□ 18-25	26-35	□ 36-45	
4. Current Employment Status			
Full-time	Part-time	☐ Unemployed	
5. No of Children			
□ 1-2	□ 3-4	5 or more	
6. Do you have health insurance?			
□Yes	□ No		
7. Type of Health Insurance			
□ Private	Public	□ None	
8. Currently Using Contract	ception		
□Yes	\square No		

Interview Questions

- 9. Can you tell me about yourself and your experience with giving birth here in the United States?
- 10. Can you describe to me an instance when you went to a postpartum family planning counseling visit?
- 11. What was your experience with your healthcare provider during postpartum family planning counseling?
- 12. How did your status as an African immigrant contribute to the service you experienced?
- 13. How did the information received during postpartum family planning counseling influence your decision to use contraception?
- 14. Before receiving postpartum family planning counseling in the United States, did you know about family planning methods before?
- 15. Describe your experience with the healthcare system in getting any form of contraception.
- 16. Did this experience affect the type of contraception you decided to use?
- 17. How, if at all, did your type of insurance contribute to your choice of contraceptive methods?
- 18. Is there anything else you would like to share?

I would like to thank you for participation. Do you have any questions?